CONTEXTUAL PERSONALIZED RECOMMENDATION TECHNIQUE FOR ACADEMIC EVENT SELECTION

PARUKUTTY RAMAN

FACULTY OF COMPUTER SCIENCE AND INFORMATION TECHNOLOGY UNIVERSITY OF MALAYA KUALA LUMPUR

2018

CONTEXTUAL PERSONALIZED RECOMMENDATION TECHNIQUE FOR ACADEMIC EVENT SELECTION

PARUKUTTY RAMAN

THESIS SUBMITTED IN FULFILMENT OF THE REQUIREMENTS FOR THE DEGREE OF MASTER

FACULTY OF COMPUTER SCIENCE & INFORMATION TECHNOLOGY UNIVERSITY OF MALAYA KUALA LUMPUR

2018

UNIVERSITY OF MALAYA ORIGINAL LITERARY WORK DECLARATION

Name of Candidate: Parukutty Raman

Matric No: WGA120008

Name of Degree: Master of Computer Science

Title of Project Paper/Research Report/Dissertation/Thesis ("this Work"):

CONTEXTUAL PERSONALIZED RECOMMENDATION TECHNIQUE FOR

ACADEMIC EVENT SELECTION

Field of Study: INFORMATION SCIENCE

I do solemnly and sincerely declare that:

- (1) I am the sole author/writer of this Work;
- (2) This Work is original;
- (3) Any use of any work in which copyright exists was done by way of fair dealing and for permitted purposes, and any excerpt or extract from or reference to or reproduction of any copyright work has been disclosed expressly and sufficiently, and the title of the Work and its authorship have been acknowledged in this Work;
- (4) I do not have any actual knowledge nor do I ought reasonably to know that the making of this work constitutes an infringement of any copyright work;
- (5) I hereby assign all and every right to the copyright of this Work to the University of Malaya ("UM"), who henceforth shall be owner of the copyright in this Work and that any reproduction or use in any form or by any means whatsoever is prohibited without the written consent of UM having been first had and obtained;
- (6) I am fully aware that if in the course of making this Work I have infringed any copyright, whether intentionally or otherwise, I may be subject to legal action or any other action as may be determined by UM.

Signature of Candidate

Date:

Subscribed and solemnly declared before,

Signature of Witness

Date:

Name:

Designation:

ABSTRACT

A recommender system in the academic domain has significant importance to researchers. Incorporating contextual information in a recommender system is vital to ensure that the recommended information is relevant and in accordance to the user's preferences. Consideration of "context" in the selection of an academic event is likely to have a profound effect on retrieving better recommendation results. Academic events can be classified as events related to any academic domain discipline, such as workshop, seminar or conference. A Recommendation System (RS), using a classical filtering approach, tends to fail when insufficient user preference information is available. The two main classical approaches are content based and collaborative. The objective of this study is to identify the most important contexts in selecting an academic event and develop a contextual personalized recommender technique for academic event selection. A survey is conducted to identify the most important contexts. Four important contexts of time, schedule, location and cost were identified and used in the technique development. Next, a tool was developed using context pre-filtering and collaborative searching techniques. The context will be pre-filtered and a search input that carries contextual data and keywords will be used to search for relevant events using a match matrix from the event database. The same events will then be checked again to see whether they have been attended by any neighbour of the user using the Top N weighted nearest neighbour technique. Contexts and keywords are explicitly given by the user. Average precision and mean average precision are used to evaluate the tool. The results will show that contextual personalized event selection techniques produce more relevant results than a technique that only uses classical approaches. This study proposes a context based personalized recommender technique to assist researchers in finding relevant academic events. The developed technique can also be used in other domains.

ABSTRAK

Sistem pengesyoran dalam domain akademik mempunyai kepentingannya yang tinggi kepada penyelidik. Maklumat kontekstual dalam sistem pengesyoran adalah penting untuk memastikan bahawa maklumat yang disyorkan adalah relevan dan mengikut keutamaan pengguna. Pertimbangan "konteks" dalam pemilihan program akademik akan memberi kesan yang positif ke atas cadangan keputusan yang lebih baik. Program akademik boleh diklasifikasikan sebagai program yang berkaitan dengan mana-mana disiplin dalam domain akademik seperti bengkel, seminar atau persidangan. Sistem pengesyoran menggunakan pendekatan penapisan klasik, yang cenderung untuk gagal apabila maklumat pengguna tidak mencukupi. Kedua-dua pendekatan klasik utama adalah berasaskan kandungan dan kerjasama. Objektif kajian ini adalah untuk mengenal pasti konteks yang penting dalam memilih program akademik dan membangunkan teknik pengesyoran peribadi kontekstual untuk pemilihan program akademik. Satu soal selidik telah dijalankan untuk mengenal pasti konteks yang penting. Empat konteks penting iaitu masa, jadual, lokasi dan kos telah dikenal pasti dan digunakan dalam pembangunan teknik. Seterusnya, satu alatan telah dibangunkan menggunakan konteks pra-penapisan dan teknik pencarian kerjasama. Konteks-konteks tersebut akan pra-ditapis dan input carian yang membawa data mengikut konteks dan kata kunci akan digunakan untuk mencari program yang berkaitan dengan menggunakan matriks padanan dari data yang telah disimpan. Program yang sama akan disemak semula untuk mengenal pasti sama ada program akademik tersebut dihadiri oleh mana-mana jiran pengguna menggunakan wajaran teknik jiran terdekat "Top N". Konteks dan kata kunci akan diberikan oleh pengguna. Purata ketepatan dan ketepatan purata min digunakan untuk menilai teknik ini. Keputusan menunjukkan bahawa teknik konteks pemilihan program akademik peribadi menghasilkan keputusan yang lebih relevan daripada teknik yang hanya menggunakan pendekatan klasik. Kajian ini mencadangkan teknik pengesyoran peribadi berasaskan konteks untuk membantu penyelidik mencari program akademik yang berkaitan. Teknik ini juga boleh digunakan dalam bidang-bidang lain.

TABLE OF CONTENTS

Abst	ract	iii	
Abst	rak	iv	
Tabl	e of Contents	v	
List	of Figures	ix	
List	of Tables	X	
List	of Symbols and Abbreviations	xi	
List	of Appendices	xii	
CHA	APTER 1: INTRODUCTION	13	
1.1	Introduction	13	
1.2	Research Motivation	14	
1.3	Problem Statement	15	
1.4	Research Objectives	16	
1.5	Research Questions	16	
1.6	Research Methodology	16	
1.7	Research Scope	18	
1.8	Thesis Outline	18	
CHA	APTER 2: LITERATURE REVIEW	20	
2.1	Introduction	20	
2.2	Academic Events		
2.3	Recommender System	21	
	2.3.1 Content-Based Filtering Technique	23	
	2.3.2 Collaborative Filtering Technique	24	

v

	2.3.3	Hybrid Filtering Technique	25
2.4	Contex	t-Aware Recommender System	26
	2.4.1	Contextual Taxonomy	27
	2.4.2	Contextual Filtering Techniques	28
	2.4.3	Contextual RS in other Domains	29
2.5	Acader	nic Event Recommender Systems	31
	2.5.1	Existing Tools and its Limitations	32
2.6	Summa	ary	34
CH	APTER	3: RESEARCH METHODOLOGY	35
3.1	Introdu	iction	35
3.2	Require	ement Planning Phase	35
	3.2.1	Information Gathering	36
		3.2.1.1 Sampling	36
		3.2.1.2 Questionnaire	36
		3.2.1.3 Data collection	
	3.2.2	Data Analysis	
3.3	User D	Design Phase	40
	3.3.1	Requirements Analysis	40
	3.3.2	Identification of the technique	41
	3.3.3	Development of Technique	42
3.4	Constru	uction Phase	43
	3.4.1	Development of Tool	43
3.5	Cut-Ov	ver – Testing Phase	43
	3.5.1	Evaluation of the Technique and Tool	43

3.6	Summary	44
СНА	PTER 4. TECHNIQUE AND TOOL DEVELOPMENT	46
A 1	Introduction	.
4.1	Contactual Personalized Recommander Technique	40
4.2	Contextual Personanzed Recommender Technique	40
4.3	Contextual Pre-Filtering Process	48
4.4	Proposed Technique	50
4.5	Summary	52
CHA	APTER 5: EVALUATION OF THE TECHNIQUE	53
5.1	Introduction	53
5.2	Findings	53
5.3	Experiment Result	54
5.4	Evaluation Results	57
5.5	Summary	57
CHA	APTER 6: CONCLUSION AND FUTURE WORK	59
6.1	Research Findings	59
6.2	Significance of the Study	61
6.3	Limitations and Future Work	61
6.4	Conclusion	62
REF	ERENCES	63
BIBI	LIOGRAPHY	67
Appe	endix A: SURVEY QUESTIONNAIRE	78
Appe	endix B: SURVEY RESPONSE	81

Appendix C: SURVEY RESPONSE CHARTS	
Annandiy D: DDOCD AMMING CODE	104
Appendix D. FROORAMIMINO CODE	104

university

LIST OF FIGURES

Figure 1.1 Research Methodology Phases	18
Figure 2.1 Recommender System Filtering Technique Architecture	23
Figure 2.2 Contextual Information Hierarchy	26
Figure 3.1 Important Contexts in Choosing an Academic Event	39
Figure 3.2 Overall Architecture of the Contextual Personalized Tool	42
Figure 4.1 Contextualization Process using Contextual Pre-Filtering	48
Figure 4.2 Contexts Hierarchy	48
Figure 4.3 Input, Process and Output for Proposed Technique	51
Figure 4.4 Pseudo Code of the Proposed Technique	52
Figure 5.1 Pilot Study of Technique and Tool Evaluation	54

LIST OF TABLES

Table 2.1 Contextual Taxonomy	27
Table 2.2 Comparison Chart of Existing Tools.	33
Table 3.1 Survey Questionnaire	37
Table 4.1 Example of User Preferences Distribution	49
Table 5.1 Experimental results for precision calculation	55
Table 5.2 Experimental results for recall calculation	56

LIST OF SYMBOLS AND ABBREVIATIONS

- RS : Recommender system
- CARS : Context-aware recommender system
- LCARS : Location context-aware recommender system
- CF : Collaborative Filtering
- CB : Content Based
- KNN : K Nearest Neighbour
- AP : Average Precision
- RQ : Research Question
- RO : Research Objective
- CM : Context Modelling

LIST OF APPENDICES

Appendix A: Survey Questionnaire	68
Appendix B: Survey Response	71
Appendix C: Survey Response Charts	90
Appendix D: Programming Code	95

university water

CHAPTER 1: INTRODUCTION

1.1 Introduction

The current information system which has emerged over the years has to deal with a huge amount of data, thereby leading to the delivery of massive results. Thus, one solution to this problem is recommender system that can filter and personalize data according to the needs of users (Pommeranz, Broekens, Wiggers, Brinkman & Jonker, 2012). A recommender system (RS) is an information filtering system that deals with the problem of information overload in current information systems (Pham, Kovachev, Cao, Mbogos & Klamma, 2012). One can use RS technology to predict the preference ratings of items that no user has currently rated, and/or to output a personalised item ranking that will likely be interesting to the user (Ricci, Rokach, Shapira & Kantor, 2011). It uses three main classical filtering techniques, namely content-based, collaborative-based and hybrid techniques, to filter the information and to retrieve relevant information. A content-based filtering technique uses the past history of the user to recommend an item, while a collaborative-based technique looks for a neighbour that has similar interests with the user to recommend items. A hybrid filtering technique is a mixture of the two techniques mentioned above. Theoretically, a collaborative filtering technique uses two main methods, which are model-based and memory-based, to filter.

A recent trend is the use of a context-aware RS as a component in a computing paradigm (Suvarna & Menezes, 2016). This use has been rapidly increasing in many domains over the years to improve the performance of the RS. According to Dey (2001), context is defined as any information that can be utilised in the characterisation of the situation of an entity, which can either be a place, person, or object. Furthermore, this characterisation is considered to possess a relevance to the interaction between an application and the user, including the actual applications and the user themselves. Ranganathan and Campbell (2003) state that context refers to any information about the objects, circumstances, or conditions that surround a user and that is considered to have relevance to the interaction between the ubiquitous computing environment and the user. The inclusion of the context in the recommendation engine may improve the performance under certain circumstances (G Adomavicius & Tuzhilin, 2005).

Academic event refers to events that academicians academically organise not just to share ideas and knowledge but also to collaborate (Pham et al., 2012). It refers to workshops, seminars, and other educational events that have one or more subject matter experts sharing information mainly via discussion and lecture. Recommendations with regard to the selection of academic events play an important role in the retrieval of the most important events for students to attend. Furthermore, the context of students in the selection of academic events is important in ensuring that the most relevant events are recommended to students for them to attend. In this study, the important contexts concerning the student's choice of an academic event were identified, followed by the development and evaluation of a tool incorporating contextual elements that are relevant to the evaluation matrix.

1.2 Research Motivation

Recommendations for academic events should be based on user preferences. This will ensure that students find the most relevant academic events according to their preferences, rather than making random selections. It is vital for students to find the most relevant events to gain the appropriate knowledge for their studies and to improve their skillset. While a substantial amount of research has already been performed in the area of recommender systems, most of the existing classical approaches focus on recommending the most relevant items to users without taking into account any additional contextual information, such as time, location, and interest (Zhou, Luo, & Xu, 2012).

In order to solve some of the problems encountered in classical recommender techniques, a contextual RS has attracted the attention of academicians (Liu, Haohan, Zhang, Hui & He, 2015). Contextual data help to produce better recommendations by taking into count user preferences and other contexts that may affect a user's choice of an item. In the selection of academic events, each student has his/her own preferences, and it is essential that these preferences (contexts), which affect the selection of academic events, be taken into considerations. The consideration of contexts in event selection will have a positive impact on the recommendation result.

1.3 Problem Statement

Each student may have their own preferences for attending an academic event, and this has to be identified accordingly. Nonetheless, students find it hard to determine the academic event that is the most relevant (Xia et al., 2013). When the wrong academic event is selected, the student will be not able to obtain the correct knowledge or skill that is relevant to their field of study and interest. The creation of recommendations for users in an academic domain to cater to their needs and tasks requires a deeper analysis of the contextual information that affects decision-making in this domain (Champiri, Shahamiri & Salim, 2015). Similarly, an RS that uses classical filtering approaches tends to fail when little knowledge about the user is known, and contextual factors in the academic domain have not been explored much over the years.

1.4 Research Objectives

The primary goal of this study was to retrieve more relevant academic events for students to attend by taking into consideration the contextual data that have an impact on the selection of academic events. The contextual personalized recommender technique was aimed at achieving the following research objectives:

- 1. To identify the user context for the selection of academic events.
- 2. To develop a contextual personalised recommender technique.
- 3. To evaluate the developed technique.

1.5 Research Questions

The research questions for this study were aimed at determining the relevance of the technique used to find academic events, as described below. The research questions were based on the research objectives mentioned below.

Objective 1: To identify the user context for the selection of academic events Research Question: What are the important contexts for the selection of academic events?

Objective 2: To develop a contextual personalized recommender technique Research Question: How can a system be developed using a contextual personalised technique?

Objective 3: To evaluate the developed technique Research Question: How can the developed technique be evaluated?

1.6 Research Methodology

Three main methodological phases were involved in this study, namely, the identification of requirements, the development of a technique and tool, and the

evaluation phase. The technique and tool development process make use of the Rapid Application Development (RAD) methodology. This methodology is used to develop the tool with the use of its four phases - requirement planning phase, construction phase, user design phase, and cutover phase. The methodology helped this study to quickly adapt to changes and to develop the tool and technique at a faster pace.

In the first phase, the requirement planning focused on the gathering of information using online surveys, followed by data extraction to derive the relevant information for this study. In the first phase of the RAD, the information provided by the earlier phase was analysed to help in the development of the technique. In the user design phase, all the information was further compiled for integration into a design. The development of the technique was carried out in this phase.

In the construction phase, the developed technique was incorporated into the tool development. All the developments were completed in this phase, where the next phase, known as the cutover phase, involved the testing of the tool. The functionality of the tool and technique tested and evaluated in this study using precision and recall measurements. The evaluation and comparison of this tool against the classical approach were performed in this phase as well. Detailed evaluation is explained later in chapter 5.



Figure 1.1 Research Methodology Phases

1.7 Research Scope

This study focused mainly on students of the University of Malaya, and covered academic events that were conducted for them within or outside the university. Thus, it had certain limitations with regard to the primary goals of this study.

- It only covered the selection of academic events by students within the University of Malaya
- 2. The retrieval of events was based on the academic events conducted at the University of Malaya.
- 3. Only students of the University of Malaya were chosen for the survey.

1.8 Thesis Outline

This study is comprised of six chapters. Chapter 1 presents an overall discussion of the recommender system (RS), the context-aware RS, and the academic event selection. It further goes through the research objectives, questions, scope and research motivation. Chapter 2 continues with the literature on previous work with regard to recommender systems, contextual recommender systems, and contextual recommender systems for the

selection of academic events. Furthermore, it also discusses the existing tools and explores their limitations in previous studies to support the proposed technique. Chapter 3 focuses fully on the methodology used to gather the information, and to develop and evaluate the technique. From there, Chapter 4 explains how the tool and technique were developed. The evaluation of the technique is clearly explained in Chapter 5, with a comparison of the experimental data and the results in order to discuss the technique measurements. The data from the experiment were examined and presented to analyse the contextualized technique. The last chapter, Chapter 6, presents the conclusion, a deliberation about the contribution of this study, and future work.

CHAPTER 2: LITERATURE REVIEW

2.1 Introduction

Academic events are basically events organized by stakeholders in the academic field. The primary purpose of these events is to share knowledge and to collaborate with other students. Academic events can be seminars, workshops, technical talks, upskill programs, and other academic-related programs. Academic events play an important role in providing information about study strategies, tools or skills. Undoubtedly, they help students to enhance their skills academically and to apply the information in their academic domain. Furthermore, academic events also enhance academic profiles. For example, students joining thesis writing workshops tend to have an edge over those who do not because they learn skills about how to write an effective thesis. As this study focused more on a recommendation technique to get students to see which the more relevant academic events, recommender systems, and contextual recommender systems. Additionally, it also discusses the limitations of previous works and the existing tools that are available for finding academic events.

2.2 Academic Events

As mentioned above, academic events are crucial for academicians to share, learn and apply the knowledge gained to enhance, improve their skills and to collaborate effectively with other students (Xia et al., 2013). Thus, selecting an academic event is very important because each student has his/her own field of study and acquires a different skill set for self-improvement (Xia et al., 2013). In order to get the right knowledge, it is vital for students, especially postgraduates, to select the most accurate or relevant academic event academ

where they will be lacking in the skill set and strategies necessary for their studies. On the other hand, the large amount of data being recommended to students may cause them to miss the relevant academic events. Students find it difficult to do the selection manually by reviewing the events one by one to see if it fits their preference and also whether it is related to their academic domain.

2.3 Recommender System

A recommender system (RS) is one way to help users deal with the flood of information (Huynh, Luong & Hoang, 2012). With the large amount of data being distributed across the Internet for any field, especially the academic field, it is crucial to have an information filtering technique for a more organized retrieval of suitable information. As previously mentioned, RS helps manage vast amounts of data. It refers to a set of technologies that are utilised in sorting goods or information from a complete information source that a user might find interesting. For example, a well-known website like Amazon.com may recommend a huge number of items to users, where it would be impossible to show all of them at a glance to the user. So, when selecting a set of items to be shown to a user, it is very important to ensure that it is something that is preferred by or is of interest of the user. Different individuals have different preferences. Hence, the system must be able to identify specific or almost similar items that are of interest to the user and are more likely to be relevant to all users. A recommender system does just that. It creates a list of items that are likely to be of interest to each individual user. In summary, "any system that produces individualized recommendations as an output or has the effect of guiding the user in a personalized way" can be defined as an RS (Sieg, Mobasher & Burke, 2007).

RS is deemed as a system that is able to provide personalized recommendations through different types of algorithms (usually termed 'filtering'). RS also can be described as a predicting system that work based on the interaction between a consumer (and business) with particular type of product and services (Bobadilla, Ortega, Hernando, Gutiérrez, 2013). It captures the taste of the specific customer based on their actions (comments, number of views, ratings) towards a particular product.

The main two units in the RS are the user and product (Madhusree, 2016). The users may share their view, opinion and experience about a product by utilizing the RS system well. Thus they able to receive recommendations about the new similar items based on their interest from the system as well. The recommendation is actually the unique preference of each customer towards the products in the system.

A successful recommender system has several conditions to be met. The primary element is requiring to contain with many items and offers under each domain. Next, the system requires to provide taste based choices to the customer and not limited for one solution. Subsequently, the system requires to equip interactive platform for the customer to review and express their view or experience about the product. Forth, the products under each domain must be homogenous so that all of the products can be covered or displayed under a taste.

An RS helps to retrieve relevant items using three main filtering approaches, namely collaborative filtering (CF), content-based (CB) and hybrid filtering. CF and CB are the classical approaches used in RS. CF performs recommendations based on the similar interests of other users, while the content-based approach uses the user's history or the features of items (Klamma, Cuong, & Cao, 2009). Hybrid approaches achieve a better

recommendation result by combining CF and content based with few other techniques. The figure below shows the architecture of recommender system techniques and the collaborative filtering categories.



Figure 2.1 Recommender System Filtering Technique Architecture

2.3.1 Content-Based Filtering Technique

The concealed principle of the content-based (CB) filtering technique is where recommendation systems analyse item descriptions to identify those items that are of particular interest to the user (Pazzani & Billsus, 2007). A classical CB recommendation system recommends items to a user based completely on the past history of that particular user, and a profile is constructed by analysing the items that are of interest or have been purchased by this user. Examples of such systems are InfoFinder, NewsWeeder and many

more. A classical CB system has several specific drawbacks. The use of only the past history of the user may lead to cold start problems when the user does not have any past history to be analysed and to recommend items (Boehmer, Jung, & Wash, 2015).

2.3.2 Collaborative Filtering Technique

Collaborative filtering (CF) refers to a RS technique that is commonly utilised in commercial applications. Recommendations are provided by CF based on the previous preferences of the user and the opinions that other users who possess similar preferences have (Breese, 1998). User preferences can be obtained through implicit and explicit methods, where an implicit method interprets a user's behaviour by using the user's history of purchases or data browsing and so on. An explicit method uses the user's rating for an item or the explicit information given by the user to recommend items. As shown in Figure 2.1 above, a CF filtering algorithm can be categorized as memory-based or algorithms model-based. Memory-based collaborative filtering generate the recommendations by operating on the entire user-item database; model-based collaborative filtering algorithms on the other hand make use of the user database for learning a model that it can then use for a recommendation. A user-based CF mainly uses the similarity between users (neighbours) to make recommendations, where it works entirely on a user-item-based matrix. This system uses a memory-based algorithm such as the KNN algorithm. According to Zohreh (2014), among the classical filtering methods used in RS, the usage of CF is high across the domains.

The CF approach is in contrast to the content-based (CB) technique, where it uses the similarity of the user's likes to predict and recommend items. For this, each user will have a "nearest neighbour" with whom this user has the strongest similarity of purchasing history or search history. The strongest correlation helps to choose the neighbour. If there

is an unseen item score, it will use a combination of items to identify the nearest neighbour. It can be said that the classical collaborative method was solely developed on the basis of similarities to other users with the same interest. Examples of systems that are implementing this approach include the Bellcore video recommender, GroupLens, and many more. It has been observed that the classical CF method solves the drawbacks found in the CB method. The cold start problem can basically be identified, where the history of other users can be used to recommend items for those who do not have their own past history. However, the CF has its own shortcomings. It will be difficult for a newer item that has been added to the database to be recommended to a user until and unless someone has searched or purchased it, and then only will it appear to other users. Therefore, there may be sparseness issues if the number of users is relatively small compared to the volume of information within the system. Furthermore, a database that is very large or rapidly changing causes the coverage of ratings to become very sparse, which in turn scatters the gathering of recommendable items (Salman, 2016). Secondly, not all users have the same interest or like what others see. Hence, this will lead to poor recommendation results (Ricci, Bontcheva, Conlan, Lawless, 2015).

2.3.3 Hybrid Filtering Technique

The hybrid filtering technique combines both the content-based and collaborative techniques to provide recommendations (Liu, Haohan et al., 2015). A hybrid of the recommendation algorithm integrates other recommendation algorithms into one to recommend items to a user. Collaborative filtering (CF) and content-based (CB) technique algorithms are often combined to create a hybrid recommender system to solve the limitations found in both the CB and CF methods (Salman, 2016).

2.4 Context-Aware Recommender System

In dealing with the limitations of classical recommender methods, a context-aware recommender system (CARS) has captured the attention of academicians (Liu, Haohan et al., 2015). CARS is widely used in many domains and has been undertaken by many researchers over the years. Thus, the information for identifying the contexts used in CARS is indispensable and is key to understanding CARS in multi domains, which will eventually help in developing the technique. Papers on CARS have been identified, and the contexts used in each paper have been extracted to construct the contextual information. Using the review done by Champiri, Shahamiri & Salim (2015) as a reference, a hierarchy of contextual information was constructed. Most of the contexts used in CARS come under any of these categories.



Figure 2.2 Contextual Information Hierarchy

2.4.1 Contextual Taxonomy

Past studies of CARS were reviewed across multiple domains to explore the contextual information affecting efficiency of recommendations. The contextual information was then sorted into each category, as shown in Figure 2.2, to construct a detailed contextual taxonomy. According to Table 2.1, the contextual conditions of the user and environment are mostly used in CARS. A common 'context' of interest in many recommendation systems is location. Other contexts, such as documents and citations, are also used in the academic domain to be incorporated in recommender systems to improve the results.

Contextual Type	Features	Conditions	References
	Profile	Name, Age, Gender, Email, Occupation, Contact No., Identity No., Fax No., Address, Marital Status, Education, Income, Faculty, etc.	(Guo & Lu, 2015; Herzog & Wolfgang, 2016; Kanetkar, Nayak, Swamy & Bhatia, 2014; M. Li, Sagl, Mburu & Fan, 2016; Miyazawa, Yamamoto & Kawabe, 2013; Park, Yoo & Cho, 2006; Uddin, Banerjee & Lee, 2016; Xia et al., 2013; Yin et al., 2013)
	Preferences	Session Preferences, Domain Preferences, Speaker Preferences, Topic Preferences, Genre Preferences, Short/Long time Preferences, Budget Preferences, etc.	(Etaati & Sundaram, 2015; Gupta & Singh, 2013; Herzog & Wolfgang, 2016; Kanetkar et al., 2014; Uddin et al., 2016; Xia et al., 2013; Yin et al., 2013)
USER	Behaviour	Very keen, just seeking information, potential buyer, dispute customer, mental state, assumptions, learning personalities, health situation, user searching behaviour, user browsing behaviour, etc.	(Bourkoukou, Bachari & Adnani, 2016; Dehghani, Afshar, Jamali & Nematbakhsh, 2011; Etaati & Sundaram, 2015; Guo & Lu, 2015; Li et al., 2016; Mobasher, Dai, Luo & Nakagawa, 2002; Sarkaleh, 2012; Sieg et al., 2007; Uddin et al., 2016; Yin et al., 2013)
50	Types	Student, researcher, postgraduate, graduate, auditor, user, purchaser, traveller, tour agent, musician, listener, etc.	(Dehghani et al., 2011; Etaati & Sundaram, 2015; Herzog & Wolfgang, 2016; Miyazawa et al., 2013; Sieg et al., 2007; Xia et al., 2013)
	Knowledge	Very Basic, Basic, Proficiency, Specialty, Expertise	(Dehghani et al., 2011; Li et al., 2014)
	Social Relationship	Collaboration with other users, work team, co-author relationships & connections between scholars, colleagues' work, Google Wave, Blog information/posts, friends, companionship, family, etc.	(Dehghani et al., 2011; Herzog & Wolfgang, 2016; Li et al., 2014; M. C. Pham et al., 2012; X. H. Pham, Nguyen, Jung, & Nguyen, 2014; Singh, Shubhankar, & Pudi, 2012; Uddin et al., 2016; Zheng, 2016)
ENVIRON MENT	Location	User location, item location, services location, longitude, latitude	(Etaati & Sundaram, 2015; Gupta & Singh, 2013; Herzog & Wolfgang, 2016; Kanetkar et al., 2014; Madadipouya, 2015; Narayanan & Cherukuri, 2016; Pham et al., 2012; X. H. Pham, Jung, Vu

Table 2.1 Contextual	Taxonomy

			& Park, 2014; Sarkaleh, 2012; Suvarna & Menezes, 2016; Tiwari & Kaushik, 2014; Tri Nguyen & Jung, 2015; Véras, Pruděncio, Ferraz, Bispo & Prota, 2016; Wu, Liu, Wang & Tan, 2016; Xia et al., 2013; Yang, Cheng, & Dia, 2008; Yin et
	Time	Weekday/Weekend, Daytime/Night, Month, Year	al., 2013) (Herzog & Wolfgang, 2016; Li et al., 2016; Miyazawa et al., 2013; Park et al., 2006; Pham et al., 2012; X. H. Pham, Jung, Vu, & Park, 2014; Singh et al., 2012; Véras et al., 2016; Xia et al., 2013; Zheng, 2016)
	Weather	Rainy, sunny, cloudy	(Etaati & Sundaram, 2015; Park et al., 2006; Wu et al., 2016)
	Ambience	Temperature, humidity, sensor, noise, illumination, social ceremonies, etc.	(Etaati & Sundaram, 2015; Park et al., 2006)
	Documents	Bibliographies between papers	(Pham et al., 2012)
	Citations	Citations between papers.	(Pham et al., 2012)
OTHERS	Music Styles	Jazz, Melody, Pop, Classical	(Park, Yoo, & Cho, 2006)
	Devices	Mobility, Mobile features	(Gui et al., 2009; Herzog & Wolfgang, 2016; Sarkaleh, 2012)

2.4.2 Contextual Filtering Techniques

Adomavicius & Tuzhilin (2008; 2011) described three types of contextual representational approaches, namely contextual pre-filtering, post-filtering and contextual modelling methods as follows:

A. Contextual pre-filtering is when the contextual data is pre-processed or filtered by removing the irrelevant ratings before they are used for computing recommendations with standard (non-contextual) methods.

B. Contextual post-filtering uses a post-contextual filtering process after noncontextual recommendation methods have been applied to the data.

C. Contextual modelling (CM) assumes that the contextual information is used within the recommendation-generating algorithms together with the user and item data.

2.4.3 Contextual RS in other Domains

As described above, contextualization is widely used in many domains such as movie recommendations, e-commerce, tourism, and book recommendations. Each of these domains works differently in contextualization because each has its unique contexts based on the user, environment and other factors. The contextual data used in movie recommendations cannot be used in book recommendations and vice versa. This is due to the fact that the decision factor in the selection of a movie and in the selection of a book may vary in terms of the domain. The preferences of the users also vary individually under the same domain. For example, if Ali and Amin are looking for movie recommendations, it does not mean that both of them have the same interest. Ali may like horror movies and Amin might prefer comedies. At the same time, both also might have the same preference to watch movies at night. Hence, contextualized filtering techniques are used here to gather the preferences of these people separately and, at the same time, to match their similar preferences to provide better recommendation results.

The emergence of the RS concept of "context" in computer science as well as other related fields has a significant influence on the development of RS. Slowly but gradually, contextual information is gaining the attention of recommender systems. Ranking models can be estimated and these models may be constructed using context as they include marginal and additional data about defined items and users. Contextual data are considered to be a vital source of precision in RSs (Baltrunas, 2008). Furthermore, several studies have shown that the performance of RS can be improved by utilising contextual data (Adomavicius & Tuzhilin, 2005; Gediminas Adomavicius & Tuzhilin, 2008; Panniello & Gorgoglione, 2010).

In this study, CARS from a previous work was studied to understand how contexts work across domains. The output was discussed in a previous session. According to Veras, Prota, Bispo, Prudencio, Ferraz (2015), the study focused on a combined cross domain context-aware model for the recommendation of television shows and books. They used contextual pre- and post-filtering techniques to recommend the contextualized data. The data were clustered according to similarity across the domain before the contextualization took place. The study was limited to only one context, namely, location. Later, Dehghani (2011) did a study on a multi-layer contextual model for digital libraries which uses multiple contexts without considering the location. Using too many contexts also creates the problem of sparseness in an RS. Other than that, a location-aware RS for movies was developed in 2015 by Madadipouya, where a CF Pearson Correlation algorithm was incorporated with contextualized data to provide more effective movie recommendations to users. Location is a context that can be used across many domains, including the academic domain. Tourism is also one of the domains where contexts are being widely used, mainly because location is one of the most important contexts for tourism recommendation systems. A tourism RS based on location, mobile devices and user features was developed by Sarkaleh in 2012 to give better recommendation results. A content-based image retrieval RS was explored by Miyazawa in 2013 using implicit attributes. Another content-based music RS developed in 2006 using a Fuzzy Bayesian technique also uses mood as its context to recommend music but it is limited to contentbased and also implicit attributes. It uses a dynamic context like mood and only a contentbased approach for recommendations, which may lead to cold start issues.

2.5 Academic Event Recommender Systems

Quite a number of studies have been done in the academic domain, such as a scientific paper recommender system using a layered approach (Manouselis & Verbert, 2013). In addition, a social awareness recommender system using strong social ties between the presenter and the participant to recommend good presentation sessions was developed in 2014 by Asabere. It is a contextualized venue recommender system that can be used in academic event recommenders. However, there are at times good sessions occurring at two different venues at the same time, thereby causing issues with regard to which one to select. An RS using disjointed user/item sets was introduced by Hornick in 2012. However, this study derived limited recommender systems that are being explored in the academic domain, a systematic review of scholar context-aware recommender systems in 2014 by Champiri confirmed that there is less incorporation of contextual data in recommender systems in the academic domain compared to other domains like e-commerce, movie recommendations, tourism and so on.

CB and CF techniques are considered as classical approaches to recommender systems, where the past history of the user or similarities in the user community are used to recommend items without taking into consideration user preferences and also the context of the environment of a user. According to Qiao, Zhang, Cao, Zhou & Guo (2014) in their study to improve collaborative recommender systems, those recommender systems that use classical approaches tend to fail when little knowledge about the user is known or no one has a similar interest to the user.

2.5.1 Existing Tools and its Limitations

There are a number of academic event recommendation tools available. Tools and methodologies developed for academic event management still faces issues (Klamma et al., 2009). On the other hand, this study focused on a personalized academic event recommender for postgraduate students from the UM, so that it concentrated on those events that are academic-based and that are also according to user preferences. There are many existing tools for recommending an event, however the most related tools are Acadevent (AcadEvent, 2017), Eventbrite and Allconferences (AllConferences, 2000).

One of the existing tools is Acadevent (AcadEvent, 2017). Acadevent uses classical (content and collaborative) filtering techniques to recommend events, and it does not consider user preferences and other environmental contexts. Users who register will be asked to provide only basic information, and they will able to use the tool to search for events. There are chances that the events recommended are not those that are preferred by the user since the user only uses a keyword to search. All the events that match the keyword will be retrieved regardless of whether it is based on the user's preferences. This also causes the tool to retrieve a huge amount of data, and the user will have to manually review each event before choosing the most relevant one.

Another tool is Eventbrite (Eventbrite, 2006), which carries out some filtering with regard to the location and type of event. However, this tool does not take into consideration the preferences of different users. The filtered categories are general, and a large number of events are produced. This may lead to sparseness, where the user may need to spend extra hours to retrieve the event one by one before deciding if it is relevant or irrelevant according to his/her preferences.

Another existing academic event recommendation tool is Allconferences (AllConferences, 2000). Allconferences is a tool to search for conferences all over the world. However, the recommendations in this tool is very wide. Even though filtering can be done in this tool, it does not take into consideration of the user contexts. Thus will produce a vast results and led to sparseness. It produces a long list of event according to keyword and locations, but user still need to go and filter the results manually or check one by one, to see the more relevant academic events for them. This tool also does not include workshops, seminars and other small scaled academic events and only focus on big conferences all over the world. Table 2.2 is the comparison summary of all three tools explained above.

Tools	EventBrite	AcadEvent	AllConferences
Saarah	Keywords and	Tagging and	Keywords and
Scarch	venue	Keywords	location
User Context	No	No	No
Web Enable	Yes	Yes	Yes
Search by Domain /	Yes	No	Yes
Categories			
Search by User	No	No	No
Preferences			
Notification of	Yes	No	Yes
Upcoming event			
Search by Location	Yes	No	Yes
Search by Time	No	No	Yes
	Sparseness and	Did not give	Sparseness
Limitation	did not give	consideration to	
Liiiitatioii	consideration to	user preference	
	user preference		

Table 2.2 Comparison Chart of Existing Tools.

2.6 Summary

In summary, based on previous studies, it can be clearly seen that context-aware recommendations have not been explored much in academic event recommender systems compared to other domains and classical approaches that tend to fail without having the contexts of the user and the environment to provide an effective recommendation. Hence this study focused on a contextual personalized academic event recommender technique by taking into consideration the contexts of user preferences and location. Based on literature review, the user context and location plays vital role in deriving more relevant recommendations. Hence, later in the tool development, user contexts and location are given more important compared to other contextual information to get more relevant result.

Moreover, the collaborative filtering technique is chosen as the filtering technique as this study more focused on user and item based recommendation and the most relevant filtering is Collaborative filtering (CF). Nevertheless, CF alone is not good due to the limitations of CF where not all users have the same interest or like what others see and this may led to poor recommendations. This is where contextual data proposed to be incorporated with CF technique to produce better result of recommendation and the evaluation of this study of classical approach (CF technique) and contextual approach (CF + context) will be producing the result of the comparison.

CHAPTER 3: RESEARCH METHODOLOGY

3.1 Introduction

This chapter discusses the methodology of the study, which was implemented using rapid application development (RAD) methodology. Basically, RAD methodology is used for fast system development, and it is a tighter fit between user requirements and system specifications. It consists of four main phases, namely requirement planning phase, user design phase, construction phase, and cutover phase. RAD is known to provide the ability to swiftly develop an application or technique, and it is flexible in terms of modifications or changes to requirements without leading to a repetitive development environment. Dennis et al (2000) and Martin (1991) defined RAD as a development life cycle that is meant to offer higher quality results and faster development compared to the traditional life cycle. In the fourth phase of this study, called the cut-over phase, the technique and tool are evaluated and the results are constructed.

3.2 Requirement Planning Phase

The requirement planning phase is a phase in which the requirements of this study are gathered and pre-processed prior to the development of the technique and tool. In this phase, the requirements are identified. Since the main objective of this study was to develop a contextual technique for academic events, it was essential to identify the most important contexts of academic events that should be contextualized and used for the technique development. It was crucial to identify the contexts as these would carry their own weightage during the filtering technique, and have an impact on the resulting technique and tool.
3.2.1 Information Gathering

Information gathering can be done in many ways such as by observation, behaviour tracking, focus groups, self-assessments, interviews, surveys, case studies and existing data. Academic events are events that are mostly attended by postgraduate students to develop certain skills and to collaborate with other participants (Pham, Kovachev, Cao, Mbogos, & Klamma, 2012). Hence, postgraduates were selected as the target group for this study.

3.2.1.1 Sampling

As mentioned in the above section, academic events are more related to postgraduates where they need to collaborate ideas with other researchers for better research outcome. Therefore, homogeneous sampling is chosen where the people from similar background and experience is chosen as a sampling for this study to reduce variation (Patton, 2001). This is sampling widely used when we use a focus group, in this case the focus group are the postgraduate of University Malaya, as the research scope is academic event conducted by University of Malaya and it is limited to students who study in University of Malaya. A questionnaire (survey) was distributed to the postgraduates. The questions were mainly based on the criteria for the selection of academic events and the most important elements that postgraduates look for when selecting an academic event. A set of questions to better understand the academic event selection pattern were listed out for the students to answer.

3.2.1.2 Questionnaire

The aim of this survey was to understand the most important elements (contexts) affecting a student's decision in choosing an academic event. This survey questionnaire consisted of 11 questions in five sections, as given below (Refer Appendix A):

Section	Question	Total number of questions
А	Identity details (level of study and faculty)	2 questions
В	Academic events attended and what makes them attend these events	3 questions
С	What are the preferred criteria when selecting an academic event?	1 question
D	What is the preferred choice (weekend/weekday, free event/with cost, morning/afternoon/night events, within campus/outside/anywhere convenient)?	4 questions
Е	Do the above criteria affect the student's choice of an academic event?	1 question
	Total	11 questions

Table 3.1 Survey Questionnaire

The survey was conducted online by means of a Google form. This research survey used the quantitative method that focused on the respondent's answers to the following questions:

- 1. What were the criteria that affected a student's choice of past academic events?
- 2. What are the expected criteria in choosing academic events?
- 3. What are the preferred choices for the criteria that have been listed above?
- 4. Do the above criteria affect a student's choice of academic events?

In section C, the criteria is distributed and not limited to only particular criteria. Students are not limited to choose the criteria, if there is a criteria out of the listed ones, the also can write that down. This is to ensure other contexts which affect the students decision choosing an academic event captured accordingly. However in section D, this question more focused on time and location based questions. Location is said one of the most dominated context in the context aware recommender system and time together can produce a more innovative and efficient tool (Xia et al., 2013). Based on this the question

is developed to know if the time and location context influence the decision making of the students.

3.2.1.3 Data collection

This survey received a very good response from the postgraduates. These postgraduates were from different fields of study who had attended academic events prior to participating in this study. From the results, it was encouraging to see that 99% of the respondents were postgraduates because this study was mainly focused on postgraduates to identify the expected criteria for choosing an academic event. Furthermore, all the respondents completed the survey, where all 12 questions were answered by everyone. Hence, the survey results were used as the deciding factor to identify the contexts that affect a student's selection of an academic event. Moreover, the reason for recruiting respondents from different fields of study was to ensure that the criteria for attending academic events were similar across the faculties. For example, the students can be from either the accounting or business faculty, but from the results obtained, both chose similar contexts when selecting an academic event. Using this pattern of answers, it can be stated for certain that the academic event selection criteria (contexts) were common among all students. On the other hand, this study was all about developing a personalized recommender tool for academic events, and therefore, the scope of the users was also focused on postgraduates alone.

3.2.2 Data Analysis

Generally, data extraction is used to provide a description of the study in general, to obtain findings from every study, and help in the interpretation of the findings and the requirement analysis phase. From the survey, four main important contexts (cost, time, location and schedule) for the selection of academic events were identified. These four main contexts carried a weightage of 80% and above, which meant that more than three quarters of the survey participants agreed that these were the important elements in the selection of an academic event.



Figure 3.1 Important Contexts in Choosing an Academic Event

Subsequent to the survey, the most to least affecting contexts of choosing academic events captured and plotted in a chart. Based on the chart, cost was the most important context, with 92 students selecting it (86.8%), followed by 91 (85.8%) students selecting time as another important context. The other two important contexts were location and schedule, which were selected by 86 students (81.1%) and 76 students (71.7%), respectively. Four contexts (cost, time, location and schedule) were identified and denoted as C1 (cost), C2 (time), C3 (location) and C4 (schedule) accordingly in the next phase.

3.3 User Design Phase

In this phase, interactions with users with the developed technique are analysed, and the input and output from and to the users are illustrated in architectural diagrams. The detailed development of technique is explained in Chapter 4, where it covers the contextual personalised recommender technique, contextual pre-filtering technique and the proposed technique which incorporate contextual technique with collaborative filtering technique.

3.3.1 Requirements Analysis

In this phase, gathered requirements in Section 3.2 were analysed to illustrate the proposed design. The four important contexts that were identified were used to translate the requirements into procedural specifications. Basically, in the recommender system, the User (U) is recommended with an Item (I) and a Rating (R). Hence, it is represented as

R: User x Item \longrightarrow R

The requirement of incorporating contexts into the classical recommendation technique will be represented as

R: User x Item x Context \longrightarrow R

Using a contextual technique, the rating function R is represented as

 $R = D1 \times D2 \times D3 \times ... \times Dn \longrightarrow Ratings$

Here, the contextual information gathered explicitly by the users of the system will be processed and used as the contextualized data to filter the events and to recommend the most relevant academic event to the user.

3.3.2 Identification of the technique

In this study, a classical technique algorithm was chosen to incorporate the contextual information so as to provide more relevant academic events to the student. Hence, the classical algorithms were reviewed, and the most suitable algorithm to incorporate the contextual information was finalized. Two main classical approaches, namely, the content-based and collaborative filtering approaches, use some algorithms to predict items to users.

The collaborative technique (CF) technique was chosen to be incorporated with the contextual data after an analysis of the advantages and disadvantages of both the classical approaches. In an academic event recommender, it is vital to take into consideration the history of other students from the same faculty or working under the same research domain who are attending an event. Thus, the CF has an edge over the CB method in recommending better results to students. However, the issue of differences in individual interests was identified using user preference contexts, where the first level of filtering was done based on user preferences before looking out for the nearest neighbour. Although the CF uses many algorithms to find the nearest neighbour, this study focused on only the user-KNN algorithm, which will be explained further in Chapter 4.

Overall, the architecture was constructed to understand how the tool and technique worked from the point of view of both the user and the organizer. Explicit inputs from the user are captured for the contexts, and the inputs are contextualized (pre-filtered) prior to the user's search. When a user searches for events, the contextualized data together with the keyword will be presented as the search input to the databases for the events listed. These events will be based on the keyword and will be filtered using the contextualized data given. Next, using the user-KNN algorithm, the nearest neighbour of the user is identified, and the events will be filtered using the KNN algorithm prior to being displayed as the results to the user.





3.3.3 Development of Technique

In this sub-section, the well-known KNN algorithm was incorporated with the context data, which had previously been explicitly given by the user. Hence, context filtering, followed by collaborative filtering, were used to develop the technique.

3.4 Construction Phase

During the construction phase, the technique and tool were combined to make a complete contextual personalized recommender tool. The requirements of the future endusers are used as the basis for the detailed definition of the design of every completed function.

3.4.1 Development of Tool

The overall architecture of the tool, as shown in Figure 3.2, was drafted to help in the development of the tool. Three main entities were developed, namely, the D1-User, D2-Organizer and D3-Event. The input given explicitly by D1 on preferences was contextualized. D2 used the same preferences to register an event being organized by them. The databases consisted of contextualized users, ratings, feedback, and events. Contextualized users here refer to users who had already registered and saved their preferences into the system. These contextualized users would later be used by the KNN-User to find the nearest neighbour.

3.5 Cut-Over – Testing Phase

The cutover phase is more on developing the test data needed to verify the system's operational capacity and also to evaluate the system relevancy to the target audience. In this study, this phase is used to evaluate the technique and tool developed and evaluation result are explained in detail in chapter 5.

3.5.1 Evaluation of the Technique and Tool

One of the objectives of this study is to evaluate the tool and technique of contextualization. Thus, a precision and recall evaluation is used to evaluate whether the relevancy matrix of the retrieved events was according to the user's preferences. Firstly,

the baseline for comparing the system against was chosen, namely, Acadevent.com. Acadevent.com was chosen as the baseline system because it uses classical approaches to recommend events. Since this study was aimed at showing that the contextualized approach is better than the classical approach, Acadevet.com considered the best to be chosen as the benchmark for this study.

Precision was chosen in performing the evaluation towards the technique and tool because it measures exactness and determines the fraction of the relevant retrieved items out of all the items retrieved. For instance, precision is used in measuring the movie recommender systems based on the proportion of recommended movies that are good in actuality. Relevant items gain more usefulness when they are found earlier in the recommendation list. Precision takes that into account as well. Whilst recall is to measure the completeness of the relevant items retrieved where it does take into count the fraction of relevant items which is not able to be retrieved in the results. Using both the precision and recall method, the technique and tool evaluated of exactness and completeness of the relevant to classical approach.

3.6 Summary

RAD was the key on developing this entire chapter. Each phases has been given equal importance from the requirement analysis phase up to the cut-over phase. This is to ensure the technique and tool able to fulfill student academic event recommendation objective and at the same time produce a more relevant results compared to the classical approaches. The technique and tool development was only briefly explained in Chapter 3, but it is the Chapter 4, which holds the detailed process of the technique and tool. Chapter 4 constructed to be more precise on the development of technique and tool in order to able to posturize the incorporation of the contextual information into collaborative filtering technique

CHAPTER 4: TECHNIQUE AND TOOL DEVELOPMENT

4.1 Introduction

This chapter explores the technique and tool development in detail, and shows the contextual personalized recommender technique in architectural figures.

4.2 Contextual Personalized Recommender Technique

The contextual personalized technique proposed in this study was a combination of context filtering followed by classical approach filtering to provide a better recommender. The contextualized data derived from the contextual filtering were further processed with the classical approach filtering to recommend items to the user.

The reduction-based approach explained by Adomavicius & Tuzhilin (2005) is an example of pre-filtering where, prior to the application of a content-based method to recommend items, the data are first computed and pre-processed with contextual data to give rise to contextualized data for further filtering. In this study, contextual pre-filtering was chosen to pre-filter the user preferences into a set of data, and these data were contextualized prior to their application in the collaborative filtering technique (user-KNN). One of the primary reason contextual pre-filtering selected is due to contextual information uses data or data constructed prior to running the recommendation engine. To be precise, information of the context, c is contextualized before the data input processed by the recommender engine. (Adomavicius and Tuzhilin, 2008).

Whilst in contextual post filtering the contextual data is intially ignored and only conextextualized after the recommendation engine recommends data from the 2D recommender. This will led to inability to fullfill the the research objective of this study as the contextual data is the input for contextual personalised recommender technique and

proposed to contextualized prior to performing the filter. On the other hand, the contextual modelling technique is where the data is used directly to the system and use a model based approach led to more complex model. This study does not use model based technique, and it adopts memory based technique and contextual modelling technique is not suitable.

The pre-filtering approach in a contextual RS uses the data in a few phases. The data and the explicit input of contexts are combined to generate the contextualized data. The user-KNN algorithm will then be used on this contextualized data to match students from the same faculty, and the events search mechanism will start to look for the best relevant event based on the student's preferences, and those that have been attended by the nearest neighbour will also be top-listed in the recommendation results. Thus, the user can easily identify the most relevant event that matches his/her preferences and that has been registered by his/her neighbour, who could be from the same faculty or working on the same domain.



Figure 4.1 Contextualization Process using Contextual Pre-Filtering

4.3 Contextual Pre-Filtering Process

In earlier chapters, the four most important contexts were finalized for incorporation into the contextualized RS. Therefore, the contexts were denoted as C1 (cost), C2 (time), C3 (location) and C4 (schedule). Each context had its own weightage, and all of the above were contextualized based on the weightage provided. Specifically, in reference to Adomavicius (2008), one can define contextual information with a set of contextual dimensions C, with every contextual dimension C in C being definable using a set of q attributes K = (K1, ..., ..., Kn). This context possesses a hierarchical structure that is similar to majority of the context-aware profiling and recommender systems. From this, the user selection can be of any set from the hierarchy, as shown in the table below. Each user there will have a set of their own preferences.



Figure 4.2 Contexts Hierarchy

An example of context conditions is described below, where six random users were chosen from different faculties, and each one of their preferences were captured. These preferences were an illustration of the distribution of preferences between users. Hence, a set of contextual data processing for each user was created during the pre-filtering process.

Contextual Conditions	Ali	Abu	Kumar	Nisa	Ah Meng	Razak
no cost	yes		yes			
with cost		yes		yes	Yes	yes
morning	yes					
afternoon					Yes	yes
night		yes	yes	yes		
within campus	yes				0,	
outside		yes	yes	yes	Yes	yes
weekday	yes		yes	yes		
weekend		yes			Yes	yes

 Table 4.1 Example of User Preferences Distribution

According to the example of the table above, each user (Ali, Abu, Kumar, Nisa, Ah Meng and Razak) has very different preferences. Ali and Kumar prefer events without any cost, but, at the same time, both have different preferences with regard to the time of the event. Similarly, Nisa, Ah Meng and Razak do not mind attending an event with cost, but only Ah Meng and Razak prefer afternoon events, while Nisa prefers night events. This shows that each user has their own unique preferences. The contexts were denoted as C1, C2 and so on, whereby each condition of the context was denoted as below.

C1 (Cost) = C1A (no cost), C1B (with cost)

C2 (Time) = C2A (morning), C2B (afternoon), C2C (night)

- C3 (Location) = C3A (within campus), C3B (outside campus)
- C4 (Schedule) = C4A (weekday), C4B (weekend)

Therefore, the user will have a set of preferences as below.

Ali = $\{C1A, C2A, C3A, C4A\}$	Nisa = $\{C1B, C2C, C3B, C4A\}$
Abu = $\{C1B, C2C, C3B, C4B\}$	Ah Meng = $\{C1B, C2B, C3B, C4B\}$
Kumar = {C1A, C2C, C3B, C4A}	Razak = {C1B, C2B, C3B, C4B}

Using this set of contextual data, the contextual pre-filtering will be done prior to the incorporation with the CF method.

4.4 **Proposed Technique**

The proposed technique, as shown in the flowchart in Figure 4.3, used three entities, namely User, Organizer and Event. Thus, these entities either provided the input or the output. A user, who is known as a system user, will register himself/herself in order to use the tool. He/she will then need to fill in the basic information to be eligible to use the tool. Moreover, it is compulsory for the student (user) to key in his/her preferences on entering the system for the very first time. The preferences can later be updated by the students based on their change of preferences. The contextual pre-filtering mechanism will use the latest updated preferences to generate a set of contextualized data for the particular user, and these preferences, the database will be updated the next time the student looks out for an event.

Organizers, who are considered to be another important entity, play a role in updating the events organized by them. Organizers will also need to go through the process of registration, and will have to key in the details of the events. The information or data on the event that is given by the organizers will be saved in a different database instance. Events, contextualized users and ratings were the three databases used in this study. At the processing level, the user context and event information provided by the organizer will be matched. If there are events that match the user preferences, they will be checked for further filtering. Once they are confirmed to be in accordance with the preferences of the user, the next search technique using the KNN algorithm will look for the nearest neighbour for this user. Databases containing the basic and contextual information of the user will be searched for faculty members or researchers from the same domain to identify the list of neighbours. The identified neighbours will be searched for their attendance at events or registration for events. The events retrieved using this technique will be displayed as the recommendation results. The display will show which events were attended by which neighbours, listing the recommendations from the highest to the least according to the relevancy to the user. The pseudo code for the proposed technique is illustrated in Figure 4.4. That are the process flow used in the contextualized technique to retrieve the most relevant events.



Figure 4.3 Input, Process and Output for Proposed Technique

Input: Set KC which contain keyword and context info, U α as a user with profile and <u>ed</u> as event database

Output: Set LHN asset of recommender events results.

- 1. Begin
- 2. For Uα search query input for keyword and context, KC ∈ ed.
- 3. For each keyword or context, KC in event database, ed,
- 4. Search with KC being registered by neighbour in ed
- 5. For all highest neighbour (HN) being registered in ed,
- If (<u>HN</u> date > current date)
- 7. Add in the list of highest neighbour, $LHN \leq 20$
- 8. Recommended event based on contexts and neighbour, LHN

Figure 4.4 Pseudo Code of the Proposed Technique

4.5 Summary

In summary, this chapter explains in detail the development of the technique and tool. The incorporation of contextual elements in classical approaches is to be evaluated to prove that the proposed technique and tool has an edge over the classical approaches. Moreover as described in literature review, context is not taking into consideration into making recommendations will lead to sparsely produced data and can cause manual filtering by the user to find the most relevant data. On the other hand, using contextual data in many RS domain has showed positive and better result of the recommendation (Adomavicius & Tuzhilin, 2005; Gediminas Adomavicius & Tuzhilin, 2008; Panniello & Gorgoglione, 2010). Hence, as stated in research problem that the contextual data is not widely explored in academic domain is the main and focal reason this technique and tool is proposed to understand how contextual data can be incorporated in academic domain to produce more relevant results.

CHAPTER 5: EVALUATION OF THE TECHNIQUE

5.1 Introduction

In this chapter, the contextualized personalized recommender technique developed for academic events was evaluated and the results were analysed using the precision measurement method. The overall results of the technique evaluation were further explored to determine if the technique met the objectives of this study.

5.2 Findings

A pilot study was conducted for the evaluation, as shown in Figure 5.1, to compare the classical approach (Acadevent.com) to the contextualized approach (proposed technique for this study). A set of retrieved items, AP@4, were then checked for relevancy according to the user's preferences. The recommended events were denoted as E1, E2, and so on. The relevant items were denoted as 'R' and the irrelevant items as 'N'. If the first prediction was relevant, 1/1=1 was given as the score, while the second relevant item in the set (E.g.: In the first set, it was the third event retrieved that was relevant) was given a score of 2/3=0.66. The third, and fourth relevant items were scored using the same method.

Once all items had been checked with regard to their relevancy, the average precision was calculated using the formula for good items. Average precision (AP) is a ranked precision metric that emphasises highly ranked correct predictions (hits).

**Precision = good items recommended / all recommendations

**Average Precision = total score of the precision / number of recommended items.

TEST 1										
(Classic	al Appro	ach		Contextualized Approach					
Retrieved Events	E1	E2	E3	E4	Retrieved Events	E	1 1	E2	E3	E4
Relevance	R	N	R	N	Relevance	R	2	R	R	N
Precision	1	-	0.66	-	Precision	1		1	1	-
AP@4		0	.40		AP@4	0.75				
TEST 2 Classical Approach Contextualized Approach										
Retrieved E Events	1 E	2 E	3 E4	4 E5	Retrieved E1 E2 E3 E4 E5 Events			E5		
Relevance	R I	R F	۱ N	N	Relevance	R	R	R	R	N
Precision	1	1 1	L -	-	Precision	1	1	1	1	-
AP@5 0.60				AP@5 0.80						

Figure 5.1 Pilot Study of Technique and Tool Evaluation

5.3 Experiment Result

The table of comparison below shows the evaluation that was carried out on 20 students using both the classical approach and the contextualized approach, where the scores for each approach were calculated based on the relevancy of the recommended items. Using the pilot study, it was shown that the contextualized approach had an edge over the classical approach, where the top-listed items that were recommended were relevant compared to the classical approach.

Data were retrieved from 20 users to evaluate both the classical approach and the contextualized approach with two different keyword searches. The average precision for each user for 10 retrieved events (AP@10) was calculated, and all the average precisions for all 20 users were summed up to calculate the average.

Average		<u>(AP User 1 + AP User 2 + AP User 3 ++AP User 20)</u>
Precision	=	Tabel up of Harm
Overall		lotal no of Users

By using the above formula, the average overall precision was calculated for each keyword using the classical approach and the contextualized approach.

	Classical A	nnroach	Contextualized Approach		
Perpendent	Drasisian				
Respondent	Precision	AP@10	Precision AP@10		
	Keyword1	Keyword2	Keyword1	Keyword2	
1	0.66	0.831	0.79	0.805	
2	0.75	0.755	0.785	0.916	
3	0.66	0.733	0.831	0.732	
4	0.525	0.79	0.733	0.866	
5	0.75	0.66	0.733	0.79	
6	0.755	0.525	0.831	0.755	
7	0.5	0.785	0.916	0.5	
8	0.66	0.831	0.79	0.66	
9	0.66	0.609	0.785	0.66	
10	0.75	0.755	0.785	0.831	
11	0.66	0.5	0.916	0.733	
12	0.525	0.66	0.732	0.831	
13	0.66	0.79	0.866	0.733	
14	0.525	0.733	0.785	0.831	
15	0.66	0.66	0.831	0.733	
16	0.66	0.79	0.733	0.733	
17	0.609	0.733	0.755	0.831	
18	0.755	0.75	0.5	0.805	
19	0.5	0.66	0.66	0.916	
20	0.66	0.66	0.66	0.732	
Average Precision AP@10	0.6442	0.7105	0.77085	0.76965	

Table 5.1 Experimental results for precision calculation

Another method of evaluation used is recall to measure the relevancy of the event recommended out of all relevant events. Recall is a measure of completeness, determines the fraction of relevant items retrieved out of all relevant items. Below is the recall formula to measure the relevancy based on the recommended events to the users. Out of 15 relevant event overall, table constructed of how many relevant events retrieved with the sample of 20 users. Data constructed to show the comparison between classical approach and contextualised approach.

D 11	good items recommended
Recall =	(good item recommended + good items not recommended)

	Classical	Approach	Contextualized Approach		
Respondent	Re	call	Recall		
	Keyword1	Keyword2	Keyword1	Keyword2	
1	0.4	0.66	0.8	0.73	
2	0.46	0.66	0.46	0.73	
3	0.3	0.66	0.73	0.8	
4	0.5	0.46	0.66	0.73	
5	0.66	0.46	0.73	0.8	
6	0.2	0.66	0.8	0.8	
7	0.5	0.3	0.66	0.5	
8	0.46	0.5	0.8	0.66	
9	0.5	0.3	0.66	0.66	
10	0.3	0.46	0.8	0.66	
11	0.4	0.66	0.66	0.66	
12	0.46	0.3	0.73	0.66	
13	0.3	0.5	0.66	0.73	
14	0.5	0.3	0.66	0.73	
15	0.3	0.66	0.8	0.73	
16	0.46	0.46	0.66	0.73	
17	0.2	0.46	0.5	0.66	
18	0.4	0.66	0.8	0.66	
19	0.46	0.46	0.5	0.8	
20	0.3	0.46	0.8	0.66	
Average Recall	0.403	0.502	0.6935	0.7045	

Table 5.2 Experimental results for recall calculation

5.4 Evaluation Results

The evaluation results showed that for the first keyword, the classical approach attained an average precision of 0.6442 over the 10 retrieved items, while the same keyword search in the contextualized approach attained a slightly higher average precision of 0.77085.

The second keyword search was measured and the contextualized approach scored an average precision of 0.76965, which was higher than the average precision of 0.7105 that was scored by the classical approach. The evaluation results further confirmed that the contextualized approach was better than the classical approach. This did not mean that the classical approach did not recommend good items, but the contextualized approach recommended items that were more relevant at the top of the list compared to the classical approach, where the recommendations were distributed among the relevant and nonrelevant items.

As for the recall an average of 0.6935 achieved for keyword 1 is gained for contextual approach compared to classical approach is 0.403. Contextual approach has a higher fraction of relevancy compared to classical approach. The same goes to the second keyword search using contextual approach which register a higher fraction of relevant item which is 0.7045 compared to classical approach of 0.502.

5.5 Summary

The contextualized approach obtained the relevant items based on the preferences explicitly given by the user. Thus, when the user saw the recommended items, the top listed ones were the items that matched their preferences. Hence, the first few items that were retrieved were considered to be good recommendations based on user preferences. Classical approaches which only uses pure filtering technique derived good results. But yet, the contextual data of four element incorporation has shown a higher fraction of exactness and completeness in the recommendation results. Location, time, cost and schedule has influenced the result in general. Students explicitly given their preferences of the four context finds the recommendation result is more relevant to what they are searching compare to the classical search only by keyword. Even though classical approach does filtration to produce result, it still can be sparse as it is not what the student may be really want to be recommended on. Thus, having contextual data incorporated will help the classical approach to get more efficient and relevant results based on the user preferences.

In summary, the evaluation results showed a positive outcome from the contextualized approach, where this experiment can be extended to more users to produce the mean average for precision and recall

CHAPTER 6: CONCLUSION AND FUTURE WORK

This study investigated the preferences (contexts) of students in choosing an academic event. Numerous tools are available for the recommendation of events to students. However, the existing tools still encounter some limitations when it comes to recommending more relevant events to students. All the research questions set out earlier were answered and discussed in Chapters 2, 3, 4 and 5. This chapter discusses the findings of this study, the research contribution, its limitations and future work. According to this study, the objectives and research questions (RQs) were determined as:

Objective 1: To identify the user context for the selection of academic events.

Research Question 1: What are the important contexts in the selection of academic events?

Objective 2: To develop a contextual personalized recommender technique.

Research Question 2: How can a system be developed using a contextual personalised technique?

Objective 3: To evaluate the developed technique.

Research Question 3: How can the developed technique be evaluated?

6.1 Research Findings

The RQs and objectives for this study were revisited. The findings for each RQ are discussed below.

Objective 1: To identify the user context for the selection of academic events

Research Question 1: What are the important contexts in the selection of academic events?

Many elements affect students in their choice of academic events. This was discussed in detail in Section 3.2.1, and based on the results obtained from the questionnaire, the four most important contexts were found and were further used to develop the technique. This answered question 1. The objective 1 is achieved.

Objective 2: To develop a contextual personalized recommender technique

Research Question 2: How can a system be developed using a contextual personalised technique?

As described in Chapter 4, the contexts derived explicitly from the users were incorporated into the selected classical filtering user-KNN algorithm to process the input and retrieve recommendations based on user preferences. Contextual pre-filtering was used at the beginning of the process to contextualise the preferences into a set of inputs from each user prior to searching the inputs using the keyword. The personalization occurred when the keyword was combined with the contextual data used to search for the events. This answered question 2. The objective 2 is achieved.

Objective 3: To evaluate the developed technique

Research Question 3: How can the developed technique be evaluated?

An experiment was carried out to evaluate the proposed technique using precision, where each retrieved item was checked as to its relevance according to the preferences of the user as discussed in section 3.2.2. The results obtained were compared with the existing tools available to confirm that the contextual recommender technique was better than the classical technique. This answered question 3. The objective 3 is achieved.

6.2 Significance of the Study

This work was aimed at helping students from the academic field to find the most suitable events to participate in and to collaborate with other students. The contribution of this work was the development of a contextual personalized academic event selection recommender technique using collaborative filtering, where the user's preferences and location are taken into account to recommend more relevant academic events. Based on the literature it was proven in other RS domain that contextual data has immensely shown efficient and innovative results. Hence, this study proposed to learn and incorporate contextual data into academic event recommender to produce better result. The experimental result of this study clearly shows that incorporating contextual data will improve the recommendation result. The exactness and completeness of the relevant events retrieved were a result of contextual information incorporation into classical approach. This work will help future researchers to further explore context-aware recommender systems in the academic domain.

6.3 Limitations and Future Work

A few limitations of this study were identified, which can be further explored:

- This study only focused on the KNN algorithm. Thus, this study can be further extended to incorporate data on contexts using different CF algorithms and comparing the results.
- Sample size is small. The use of more respondents will give better average precision results for the contextual technique.

• The database for the events was limited to events happening at the University of Malaya. Importing more events will have a good impact on the results.

6.4 Conclusion

The primary focus of this study was to identify the preferences of students that affect their decision to attend academic events. The results of the investigation showed that students face difficulties in finding the relevant academic events based on their preferences due to the limitations of existing tools. Thus, this study focused on developing a contextual personalized recommendation technique that incorporates contextual information in its retrieval process. The contextual pre-filtering technique was chosen due to its suitability in helping to contextualize data prior to the searching process.

This study found that contextual information, which can also be defined as user preferences, can be mapped into recommender systems. The main significance of this study is that user preferences can be identified and incorporated into the classical recommendation technique.

A classical user-based recommendation algorithm, the user-KNN, and a context pre-filtering technique were embedded in the technique development. The contextual data were incorporated into the K-nearest neighbour (KNN) algorithm to recommend events based on user preferences.

Finally, from the experimental results obtained, it is believed that all the objectives set out at the beginning of this study have been successfully achieved.

REFERENCES

AcadEvent (2017). AcadEvent. Retrieved from acadevent.com

- Adomavicius, G., & Tuzhilin, A. (2008). Context-aware Recommender Systems. Proceedings of the 2008 ACM Conference on Recommender Systems, 335–336. https://doi.org/10.1145/1454008.1454068
- Adomavicius, G., & Tuzhilin, a. (2005). Toward the Next Generation of Recommender Systems: a Survey of the State of the Art and Possible Extensions. *IEEE Transactions on Knowledge and Data Engineering*, 17(6), 734–749. https://doi.org/10.1109/TKDE.2005.99

AllConference (2000), AllConference. Retrieved from AllConferences.com

Bobadilla J., Ortega F., Hernando A., and Gutiérrez A. (2013) Recommender systems survey, Knowledge-Based Systems, vol. 46, pp. 109–132.

- Boehmer, J., Jung, Y., & Wash, R. (2015). Recommender Systems, 1542(9), 33–36. https://doi.org/10.1017/CBO9781107415324.004
- Bourkoukou, O., Bachari, E. El, & Adnani, M. El. (2016). A Personalized E-Learning Based on Recommender System. *International Journal of Learning and Teaching*, 2(2), 99–103. https://doi.org/10.18178/ijlt.2.2.99-103
- Champiri, Z. D., Shahamiri, S. R., & Salim, S. S. B. (2015). A systematic review of scholar context-aware recommender systems. *Expert Systems with Applications*, 42(3), 1743–1758. https://doi.org/10.1016/j.eswa.2014.09.017
- Dehghani, Z., Afshar, E., Jamali, H. R., & Nematbakhsh, M. A. (2011). A multi-layer contextual model for recommender systems in digital libraries. *Aslib Proceedings*, 63(6), 555–569. https://doi.org/10.1108/00012531111187216
- Etaati, L., & Sundaram, D. (2015). Adaptive tourist recommendation system: conceptual frameworks and implementations. *Vietnam Journal of Computer Science*, *2*(2), 95–107. https://doi.org/10.1007/s40595-014-0034-5

Eventbrite (2017). Eventbrite. Retrieved from eventbrite.com

- Gui, F., Guillen, M., Rishe, N., Barreto, A., Andrian, J., & Adjouadi, M. (2009). A clientserver architecture for context-aware search application. NBiS 2009 - 12th International Conference on Network-Based Information Systems, 539–546. https://doi.org/10.1109/NBiS.2009.75
- Guo, F., & Lu, Q. (2015). Model and Its Application in e-Commerce Customer Satisfaction Management. *Discrete Dynamics in Nature and Society*, 2015, 1–11.

Gupta, A., & Singh, K. (2013). Location based personalized restaurant recommendation

system for mobile environments. *Proceedings of the 2013 International Conference on Advances in Computing, Communications and Informatics, ICACCI 2013*, 507–511. https://doi.org/10.1109/ICACCI.2013.6637223

- Herzog, D., & Wolfgang, W. (2016). Extending Content-Boosted Collaborative Filtering for Context-aware, Mobile Event Recommendations, 2(Webist), 293–303.
- Huynh, T., Luong, H., & Hoang, K. (2012). of Computer Science Publications, 226–235.
- Kanetkar, S., Nayak, A., Swamy, S., & Bhatia, G. (2014). Web-based personalized hybrid book recommendation system. 2014 International Conference on Advances in Engineering and Technology Research, ICAETR 2014, 0–4. https://doi.org/10.1109/ICAETR.2014.7012952
- Klamma, R., Cuong, P. M., & Cao, Y. (2009). {Y}ou never walk alone: {R}ecommending academic events based on social network analysis. {C}omplex {S}ciences, 657–670.
- Li, J., Xia, F., Wang, W., Chen, Z., Asabere, N. Y., & Jiang, H. (2014). ACRec: a coauthorship based random walk model for academic collaboration recommendation. *WWW 2014 Companion: Proceedings of the 23rd International Conference on World Wide Web*, 1209–1214. https://doi.org/10.1145/2567948.2579034
- Li, M., Sagl, G., Mburu, L., & Fan, H. (2016). A contextualized and personalized model to predict user interest using location-based social networks. *Computers, Environment and Urban Systems*, 58, 97–106. https://doi.org/10.1016/j.compenvurbsys.2016.03.006
- Liu, Haohan, Zhang, H., Hui, K., & He, H. (2015). Overview Of Context-aware Recommender System Research Haohan Liu, Hongli Zhang *, Kanghua Hui , Huaiqing He, (Icmra), 1218–1221.
- Madadipouya, K. (2015). A Location-Based Movie Recommender System Using Collaborative Filtering. *International Journal in Foundations of Computer Science* & *Technology*, 5(4), 13–19. https://doi.org/10.5121/ijfcst.2015.5402

Madhushree, B. (2016). A Novel Research Paper Recommendation System. International Journal of Advanced Research in Engineering and Technology, 7(1), pp. 07-16

- Miyazawa, Y., Yamamoto, Y., & Kawabe, T. (2013). Context-Aware Recommendation System Using Content Based Image Retrieval with Dynamic Context Considered. 2013 International Conference on Signal-Image Technology & Internet-Based Systems, 779–783. https://doi.org/10.1109/SITIS.2013.126
- Mobasher, B., Dai, H., Luo, T., & Nakagawa, M. (2002). Discovery and evaluation of aggregate usage profiles for Web personalization. *Data Mining and Knowledge Discovery*, 6(1), 61–82. https://doi.org/10.1023/A:1013232803866

Narayanan, M., & Cherukuri, A. K. (2016). A study and analysis of recommendation

systems for location-based social network (LBSN) with big data. *IIMB Management Review*, 28(1), 25–30. https://doi.org/10.1016/j.iimb.2016.01.001

- Panniello, U., & Gorgoglione, M. (2010). Does the recommendation task affect a CARS performance? *Proceedings of the 2010 Workshop on Context-Aware Recommender Systems*. Retrieved from http://ids.csom.umn.edu/faculty/gedas/cars2010/PannielloGorgoglione-CARS-2010.pdf
- Patton, M. Q. (2001). Qualitative evaluation and research methods (3rd ed.). Newbury Park, CA: Sage Publications.
- Park, H.-S., Yoo, J.-O., & Cho, S.-B. (2006). A context-aware music recommendation system using fuzzy bayesian networks with utility theory. *Fuzzy Systems and Knowledge Discovery*, 970–979. https://doi.org/10.1007/11881599
- Pazzani, M. J., & Billsus, D. (2007). Content-Based Recommendation Systems. *The Adaptive Web*, 4321, 325–341. https://doi.org/10.1007/978-3-540-72079-9
- Pham, M. C., Kovachev, D., Cao, Y., Mbogos, G. M., & Klamma, R. (2012). Enhancing academic event participation with context-aware and social recommendations. *Proceedings of the 2012 IEEE/ACM International Conference on Advances in Social Networks Analysis and Mining, ASONAM 2012*, 464–471. https://doi.org/10.1109/ASONAM.2012.80
- Pham, X. H., Jung, J. J., Vu, L. A., & Park, S. B. (2014). Exploiting social contexts for movie recommendation. *Malaysian Journal of Computer Science*, 27(1), 68–79.
- Pham, X. H., Nguyen, T. T., Jung, J. J., & Nguyen, N. T. (2014). <A,V>-spear: A new method for expert based recommendation systems. *Cybernetics and Systems*, 45(2), 165–179. https://doi.org/10.1080/01969722.2014.874822
- Salman, Y. A. (2016). A Proactive Multi- T ype Context- A ware Recommender System in the Environment of Internet of Things, (February).
- Sarkaleh, M. K. (2012). Designing a Tourism Recommender System Based on Location, Mobile Device and User Features in Museum. *International Journal of Managing Information Technology*, 4(2), 13–21. https://doi.org/10.5121/ijmit.2012.4202
- Sieg, A., Mobasher, B., & Burke, R. (2007). Representing Context in Web Search with Ontological User Profiles.pdf, 439–452.
- Singh, A. P., Shubhankar, K., & Pudi, V. (2012). Topic-centric recommender systems for bibliographic datasets. Lecture Notes in Computer Science (Including Subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics), 7713 LNAI(December), 689–700. https://doi.org/10.1007/978-3-642-35527-1_57
- Suvarna, A. S., & Menezes, K. W. (2016). CONTEXT-AWARE RECOMMENDER SYSTEM BY MERGING WIRELESS SENSORS AND, 176–180.

- Tiwari, S., & Kaushik, S. (2014). Information Enrichment for Tourist Spot Recommender System Using Location Aware Crowdsourcing. 2014 IEEE 15th International Conference on Mobile Data Management, 11–14. https://doi.org/10.1109/MDM.2014.59
- Tri Nguyen, T., & Jung, J. J. (2015). Exploiting geotagged resources to spatial ranking by extending HITS algorithm. *Computer Science and Information Systems*, 12(1), 185–201. https://doi.org/10.2298/CSIS141015091T
- Uddin, M. F., Banerjee, S., & Lee, J. (2016). Recommender system framework for academic choices: Personality based recommendation engine (PBRE). *Proceedings* 2016 IEEE 17th International Conference on Information Reuse and Integration, IRI 2016, 2, 476–483. https://doi.org/10.1109/IRI.2016.70
- Véras, D., Pruděncio, R., Ferraz, C., Bispo, A., & Prota, T. (2016). Context-aware techniques for cross-domain recommender systems. *Proceedings - 2015 Brazilian Conference on Intelligent Systems, BRACIS 2015*, 282–287. https://doi.org/10.1109/BRACIS.2015.42
- Véras D, Prota T, Bispo A, Prudêncio R, Ferraz C.(2015). A literature review of recommender systems in the television domain. Expert Systems with Applications. 42(22):9046–76.
- Wu, S., Liu, Q., Wang, L., & Tan, T. (2016). Contextual Operation for Recommender Systems. *IEEE Transactions on Knowledge and Data Engineering*, 28(8), 2000– 2012. https://doi.org/10.1109/TKDE.2016.2562621
- Xia, F., Asabere, N. Y., Rodrigues, J. J. P. C., Basso, F., Deonauth, N., & Wang, W. (2013). Socially-aware venue recommendation for conference participants. *Proceedings - IEEE 10th International Conference on Ubiquitous Intelligence and Computing, UIC 2013 and IEEE 10th International Conference on Autonomic and Trusted Computing, ATC 2013*, (i), 134–141. https://doi.org/10.1109/UIC-ATC.2013.81
- Yang, W. S., Cheng, H. C., & Dia, J. Ben. (2008). A location-aware recommender system for mobile shopping environments. *Expert Systems with Applications*, 34(1), 437– 445. https://doi.org/10.1016/j.eswa.2006.09.033
- Yin, H., Sun, Y., Cui, B., Chen, L., Hu, Z., & Chen, L. (2013). LCARS: a locationcontent-aware recommender system. *Kdd*, 221. https://doi.org/10.1145/2487575.2487608
- Zheng, Y. (2016). Tutorial: Context-Awareness In Information Retrieval and Recommender Systems.
- Zhou, J., Luo, T., & Xu, G. (2012). Academic recommendation on graph with dynamic transfer chain.

BIBLIOGRAPHY

- Ekstrand, M. D., Riedl, J. T., & Konstan, J. A. (2007). Collaborative Filtering Recommender Systems. *International Journal of Electronic Business*, 4321(1), 291–324. https://doi.org/10.1504/IJEB.2004.004560
- Harispe, S., Ranwez, S., Janaqi, S., & Montmain, J. (2013). Semantic measures based on RDF projections: Application to content-based recommendation systems (short paper). In *Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics)* (Vol. 8185 LNCS, pp. 606–615). https://doi.org/10.1007/978-3-642-41030-7_44
- Wang, C., & Tong, H. (2010). Research on psychological dimensions of e-commerce customer satisfaction. In *Proceedings of the International Conference on E-Business and E-Government, ICEE 2010* (pp. 2105–2108). https://doi.org/10.1109/ICEE.2010.532
- Leal, R., Costa, P. M., & Galvao, T. (2015). Discovering MARS: A Mobility Aware Recommender System. In *IEEE Conference on Intelligent Transportation Systems, Proceedings, ITSC* (Vol. 2015–October, pp. 1–6). https://doi.org/10.1109/ITSC.2015.8
- Bedi, P., & Agarwal, S. K. (2011). Aspect-oriented mobility-aware recommender system. In Proceedings of the 2011 World Congress on Information and Communication Technologies, WICT 2011 (pp. 191–196). https://doi.org/10.1109/WICT.2011.6141242
- Hammami, S., Saeed, F., Mathkour, H., & Arafah, M. A. (2017). Continuous improvement of deaf student learning outcomes based on an adaptive learning system and an Academic Advisor Agent. *Computers in Human Behavior*. https://doi.org/10.1016/j.chb.2017.07.006
- Li, X., Mi, Z., Zhang, Z., & Wu, J. (2010). A location-aware recommender system for tourism mobile commerce. In 2nd International Conference on Information Science and Engineering, ICISE2010 - Proceedings (pp. 1709–1711). https://doi.org/10.1109/ICISE.2010.5690282
- Unger, M. (2015). Latent Context-Aware Recommender Systems. AAAI Technical Report WS-99-01., 383–386. https://doi.org/10.1145/2792838.2796546
- Torrente, J., Blanco, A., Moreno-ger, P., & Fernández-manjón, B. (2012). A Contextual Bandit Algorithm for Mobile Context-Aware Recommender System. *Neural Information Processing, Lecture Notes in Computer Scienc*, 7666(November), 603–610. https://doi.org/10.1007/978-3-642-34478-7
- Haddad, M. R., Baazaoui, H., Ziou, D., & Ben Ghezala, H. (2012). Towards a new model for context-aware recommendation. In *IS'2012 2012 6th IEEE*

International Conference Intelligent Systems, Proceedings (pp. 21–27). https://doi.org/10.1109/IS.2012.6335185

- Course, A. S. (2012). Recommender Systems Evaluation Evaluation of Recommender Systems.
- Jannach, D., Zanker, M., Ge, M., & Gröning, M. (2012). Recommender Systems in Computer Science and Information Systems–A Landscape of Research. *E-Commerce and Web Technologies*, 76–87. https://doi.org/10.1007/978-3-642-32273-0_7
- Mathur, S., & Bairagee, N. (2016). A Survey Paper on Location Aware Recommender System, *6*(7), 331–335.
- Khatwani, S., & Chandak, M. B. (2017). Building Personalized and Non Personalized recommendation systems. *International Conference on Automatic Control and Dynamic Optimization Techniques*, *ICACDOT 2016*, 8(7), 623–628. https://doi.org/10.1109/ICACDOT.2016.7877661
- Herlocker, J. L., Konstan, J. A., Terveen, L. G., & Riedl, J. T. (2004). Evaluating collaborative filtering recommender systems. ACM Transactions on Information Systems, 22(1), 5–53. https://doi.org/10.1145/963770.963772
- Carmen, M., Ilarri, S., & Trillo-lado, R. (2015). Location-Aware Recommendation Systems : Where We Are and Where We Recommend to Go Categories and Subject Descriptors. *RecSys LocalRec 2015: Proceedings of the Workshop on Location-Aware Recommendations Co-Located with the 9th ACM Conference on Recommender Systems*, 1–8.
- Pham, M. C., Cao, Y., & Klamma, R. (2010). Clustering Technique for Collaborative Filtering and the Application to Venue Recommendation. *Proceedings of I-KNOW* 2010, (September), 343–354.
- Jizba, R. (2000). Measuring search effectiveness. *Creighton University Health Sciences Library and* ..., 3–5. Retrieved from http://scholar.google.com/scholar?hl=en&btnG=Search&q=intitle:Measuring+Sear ch+Effectiveness#2
- Pham, M. C., Cao, Y., Klamma, R., & Jarke, M. (2011). A Clustering Approach for Collaborative Filtering Recommendation Using Social Network Analysis. *Journal of Universal Computer Science*, *17*(4), 1–21. https://doi.org/10.3217/jucs-017-04-0583
- Shani, G., & Gunawardana, A. (2011). Evaluating Recommendation Systems. Recommender Systems Handbook, 257–297. https://doi.org/10.1007/978-0-387-85820-3_8
- Jung, J. J. (2011). Ubiquitous conference management system for mobile recommendation services based on mobilizing social networks: A case study of u-

conference. *Expert Systems with Applications*, *38*(10), 12786–12790. https://doi.org/10.1016/j.eswa.2011.04.070

- Zheng, C., E, H., Song, M., & Song, J. (2016). CMPTF: Contextual Modeling Probabilistic Tensor Factorization for recommender systems. *Neurocomputing*, 205, 141–151. https://doi.org/10.1016/j.neucom.2016.04.016
- Beel, J., Gipp, B., Langer, S., & Breitinger, C. (2014). Research Paper Recommender Systems : A Literature Survey, 20(April 2013). https://doi.org/10.1007/s00799-015-0156-0.Research
- Bari, P. (n.d.). No Title, 1–32.
- Song, L., Tekin, C., & Van Der Schaar, M. (2016). Online Learning in Large-Scale Contextual Recommender Systems. *IEEE Transactions on Services Computing*, 9(3), 433–445. https://doi.org/10.1109/TSC.2014.2365795
- Mayer, J. M., Hiltz, S. R., Barkhuus, L., Väänänen, K., Jones, Q., & Tech, C. (2016). S upporting Opportunities for Context - Aware Social Match ing : An Experience Sampling Study. *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*. https://doi.org/10.1145/2858036.2858175
- Codina, V., Ricci, F., & Ceccaroni, L. (2016). Distributional semantic pre-filtering in context-aware recommender systems. User Modeling and User-Adapted Interaction, 26(1), 1–32. https://doi.org/10.1007/s11257-015-9158-2
- Panniello, U., Hill, S., & Gorgoglione, M. (2016). Using Context for Online Customer Re-Identification-final, 64, 500–511.
- Zarka, R., Cordier, A., Egyed-Zsigmond, E., Lamontagne, L., & Mille, A. (2016). Trace-based contextual recommendations. *Expert Systems with Applications*, 64, 194–207. https://doi.org/10.1016/j.eswa.2016.07.035
- Gupta, A., & Singh, K. (2013). Location based personalized restaurant recommendation system for mobile environments. *Proceedings of the 2013 International Conference on Advances in Computing, Communications and Informatics, ICACCI* 2013, 507–511. https://doi.org/10.1109/ICACCI.2013.6637223
- Luna, V., Quintero, R., Torres, M., Moreno-Ibarra, M., Guzmán, G., & Escamilla, I. (2015). An ontology-based approach for representing the interaction process between user profile and its context for collaborative learning environments. *Computers in Human Behavior*, 51, 1387–1394. https://doi.org/10.1016/j.chb.2014.10.004
- Vo, C., Torabi, T., & Loke, S. (2005). A Survey of Context-Aware Recommendation Systems. *Homepage.Cs.Latrobe.Edu.Au*, 1–23. Retrieved from http://homepage.cs.latrobe.edu.au/ccvo/papers/16recommendation.pdf

Bari, P. (n.d.). No Title, 1–32.

- Oku, K., Nakajima, S., Miyazaki, J., Uemura, S., Kato, H., & Hattori, F. (2010). A Recommendation System Considering Users' Past/Current/Future Contexts. *Ids.Csom.Umn.Edu*, 3–7. Retrieved from http://ids.csom.umn.edu/faculty/gedas/cars2010/OkuEtAl-CARS-2010.pdf
- Roll, I., Baker, R. S., Aleven, V., McLaren, B. M., & Koedinger, K. R. (2005). An analysis of differences between preferences in real and supposed contexts. User Modeling- Springer Berlin Heidelberg, 367–376. Retrieved from http://ids.csom.umn.edu/faculty/gedas/cars2010/asohetal-cars-2010.pdf
- Frame, C., & Print, T. O. (2015). Context-aware recommender systems for learning: A survey and future challenges, 5(4), 2–3.
- Mobasher, B. (2010). Contextual user modelling for recommendation context in recommendation scenario. *Computing*. https://doi.org/10.1145/2623330.2630807
- Zhang, Y., & Wang, L. (2010). Some challenges for context-aware recommender systems. ICCSE 2010 - 5th International Conference on Computer Science and Education, Final Program and Book of Abstracts, 362–365. https://doi.org/10.1109/ICCSE.2010.5593612
- Campbell, D. R., Culley, S. J., McMahon, C. A., & Sellini, F. (2007). An approach for the capture of context-dependent document relationships extracted from Bayesian analysis of users' interactions with information. *Information Retrieval*, 10(2), 115– 141. https://doi.org/10.1007/s10791-006-9016-2
- Park, H., Yoo, J., & Cho, S. (2006). A Context-Aware Music Recommendation System. *Fuzzy Systems and Knowledge Discovery*, 4223 LNAI, 970–979. https://doi.org/10.1007/11881599
- Mobasher, B. (2011). Recommender Systems in. *Data Mining Applications with R*, 2015(September), 81–90. https://doi.org/10.1016/B978-0-12-411511-8.00005-0
- Chen, M., Yang, C., Chen, J., & Yi, P. (2013). A Method to Solve Cold-Start Problem in Recommendation System based on Social Network Sub-community and Ontology Decision Model. 3rd International Conference on Multimedia Technology (ICMT 2013) A, 159–166. Retrieved from http://www.atlantispress.com/php/download_paper.php?id=10372
- Jooa, J., Bangb, S., & Parka, G. (2016). Implementation of a Recommendation System Using Association Rules and Collaborative Filtering. *Procedia Computer Science*, 91(Itqm), 944–952. https://doi.org/10.1016/j.procs.2016.07.115
- Dhawan, S., Singh, K., & Jyoti. (2015). High Rating Recent Preferences Based Recommendation System. *Procedia Computer Science*, 70, 259–264. https://doi.org/10.1016/j.procs.2015.10.085
- Sieg, A., Mobasher, B., & Burke, R. (n.d.). Ontological User Profiles as the Context Model in Web Search.

- Wang, X., Yang, C., & Mao, S. (2017). TensorBeat: Tensor Decomposition for Monitoring Multi-Person Breathing Beats with Commodity WiFi, 0(0). https://doi.org/10.1145/0000000.0000000
- He, J., & Chu, W. W. (2010). A Social Network-Based Recommender System (SNRS), 47–74. https://doi.org/10.1007/978-1-4419-6287-4_4
- Colombo-Mendoza, L. O., Valencia-García, R., Rodríguez-González, A., Alor-Hernández, G., & Samper-Zapater, J. J. (2015). RecomMetz: A context-aware knowledge-based mobile recommender system for movie showtimes. *Expert Systems with Applications*, 42(3), 1202–1222. https://doi.org/10.1016/j.eswa.2014.09.016
- Ashley-Dejo, E., Ngwira, S., & Zuva, T. (2016). A survey of Context-Aware Recommender System and services. 2015 International Conference on Computing, Communication and Security, ICCCS 2015. https://doi.org/10.1109/CCCS.2015.7374144
- Amini, B., Ibrahim, R., Othman, M. S., & Rastegari, H. (2011). Incorporating scholar's background knowledge into recommender system for digital libraries. 2011 5th Malaysian Conference in Software Engineering, MySEC 2011, 516–523. https://doi.org/10.1109/MySEC.2011.6140721
- Yeung, K. F. (2011). A context-aware framework for personalized recommendation in mobile environments, (September).
- Abbas, A., Zhang, L., & Khan, S. U. (2015). A survey on context-aware recommender systems based on computational intelligence techniques. *Computing*, 97(7), 667– 690. https://doi.org/10.1007/s00607-015-0448-7
- Li, H., & Xu, J. (2014). Semantic Matching in Search. Foundations and Trends® in Information Retrieval (Vol. 7). https://doi.org/10.1561/1500000035
- Verbert, K., Parra, D., & Brusilovsky, P. (2016). Agents vs. Users: Visual Recommendation of Research Talks with Multiple Dimension of Relevance. ACM Trans. Interact. Intell. Syst., 6(2), 11:1–11:42. https://doi.org/10.1145/2946794
- Pham, X., Nguyen, T., Jung, J., & Hwang, D. (2014). Extending HITS Algorithm for Ranking Locations by Using Geotagged Resources. *Proceedings of the 6th International Conference, ICCCI 2014*, 332–341. Retrieved from http://link.springer.com/chapter/10.1007/978-3-319-11289-3 34
- Mazurowski, M. A. (2013). Estimating confidence of individual rating predictions in collaborative filtering recommender systems. *Expert Systems with Applications*, 40(10), 3847–3857. https://doi.org/10.1016/j.eswa.2012.12.102
- Parameswaran, A., Venetis, P., & Garcia-Molina, H. (2011). Recommendation systems with complex constraints. ACM Transactions on Information Systems, 29(4), 1–33. https://doi.org/10.1145/2037661.2037665
- Brand, M. A., & Moro, M. M. (2012). Affiliation influence on recommendation in academic social networks. *CEUR Workshop Proceedings*, 866, 230–234.
- Tejeda-Lorente, A., Porcel, C., Bernab??-Moreno, J., & Herrera-Viedma, E. (2015). REFORE: A recommender system for researchers based on bibliometrics. *Applied Soft Computing Journal*, 30, 778–791. https://doi.org/10.1016/j.asoc.2015.02.024
- Scholz, C., Illig, J., Atzmueller, M., & Stumme, G. (2014). On the predictability of talk attendance at academic conferences (Extended Abstract). *CEUR Workshop Proceedings*, 1226(September), 162–163. https://doi.org/10.1145/2631775.2631816
- Tejeda-Lorente, Á, Porcel, C., Peis, E., Sanz, R., & Herrera-Viedma, E. (2014). A quality based recommender system to disseminate information in a university digital library. *Information Sciences*, 261, 52–69. https://doi.org/10.1016/j.ins.2013.10.036
- Coates, A., & Ng, A. Y. (2012). Learning feature representations with K-means. Lecture Notes in Computer Science (Including Subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics), 7700 LECTU, 561–580. https://doi.org/10.1007/978-3-642-35289-8-30
- Ricci, F., Bontcheva, K., Conlan, O., & Lawless, S. (2015). User Modelling, Adaptation and Personalization: 23rd International Conference, UMAP 2015 Dublin, Ireland, June 29 - July 3, 2015 Proceedings. *Lecture Notes in Computer Science (Including Subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics*), 9146, 370–371. https://doi.org/10.1007/978-3-319-20267-9
- Vala Ali, R. (2014). Content-Based Recommender System for an Academic Social Network. https://doi.org/10.1109/BDCloud.2014.131
- Amini, B., Ibrahim, R., Othman, M. S., & Rastegari, H. (2011). Incorporating Scholar's Background Knowledge into Recommender System for Digital Libraries, 516–523.
- Hoang, D. T., Hwang, D., Tran, V. C., Nguyen, V. Du, & Nguyen, N. T. (2017). Academic event recommendation based on research similarity and exploring interaction between authors. 2016 IEEE International Conference on Systems, Man, and Cybernetics, SMC 2016 - Conference Proceedings, 4411–4416. https://doi.org/10.1109/SMC.2016.7844925
- Jannach, D., Lerche, L., Kamehkhosh, I., & Jugovac, M. (2015). What recommenders recommend: an analysis of recommendation biases and possible countermeasures. User Modeling and User-Adapted Interaction (Vol. 25). Springer Netherlands. https://doi.org/10.1007/s11257-015-9165-3
- Tran, V. C., Hwang, D., & Jung, J. J. (2015). Semi-supervised approach based on cooccurrence coefficient for named entity recognition on Twitter. *Proceedings of* 2015 2nd National Foundation for Science and Technology Development

Conference on Information and Computer Science, NICS 2015, 141–146. https://doi.org/10.1109/NICS.2015.7302179

- Krishnaraju, V., Mathew, S. K., & Sugumaran, V. (2016). Web personalization for user acceptance of technology: An empirical investigation of E-government services. *Information Systems Frontiers*, 18(3), 579–595. https://doi.org/10.1007/s10796-015-9550-9
- Ley, M. (2009). DBLP Some Lessons Learned *.
- Le, Q. V., & Mikolov, T. (2014). Distributed Representations of Sentences and Documents, *32*. https://doi.org/10.1145/2740908.2742760
- Jackson, M., Nelson, D., & Stirk, S. (2005). Lecture Notes in Computer Science. Lecture Notes in Computer Science. https://doi.org/10.1007/978-3-642-29066-4{_}11
- Campr, M., & Je??ek, K. (2015). Comparing semantic models for evaluating automatic document summarization. Lecture Notes in Computer Science (Including Subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics), 9302, 252–260. https://doi.org/10.1007/978-3-319-24033-6 29
- Unger, M., Bar, A., Shapira, B., & Rokach, L. (2016). Towards latent context-aware recommendation systems. *Knowledge-Based Systems*, 104, 165–178. https://doi.org/10.1016/j.knosys.2016.04.020
- Zheng, Y. (2016). Tutorial : Context in Recommender Systems.
- ligne), L. (Service. (1973). Lecture notes in artificial intelligence. Science and *Technology*. https://doi.org/10.1007/978-3-642-15314-3
- Newman, M. E. J. (2001). Scientific collaboration networks. I. Network construction and fundamental results. *Physical Review E*, 64(1), 16131. https://doi.org/10.1103/PhysRevE.64.016131
- Luong, N. T., Nguyen, T. T., Hwang, D., Lee, C. H., & Jung, J. J. (2015). Similaritybased complex publication network analytics for recommending potential collaborations. *Journal of Universal Computer Science*, *21*(6), 871–889.
- Board, E., Ferrari, D., & Gerla, M. (2009). Lecture Notes of the Institute for Computer Sciences, Social Informatics and Telecommunications Engineering. Middle East. https://doi.org/10.1007/978-3-642-27317-9_36
- Vossen, G. (2014). Big data as the new enabler in business and other intelligence. *Vietnam Journal of Computer Science*, 1(1), 3–14. https://doi.org/10.1007/s40595-013-0001-6
- Taha, K. (2012). Automatic Academic Advisor. Proceedings of the 8th IEEE International Conference on Collaborative Computing: Networking, Applications

and Worksharing, 262–268. https://doi.org/10.4108/icst.collaboratecom.2012.250338

- Khrouf, H., & Troncy, R. (2013). Hybrid event recommendation using linked data and user diversity. *Proceedings of the 7th ACM Conference on Recommender Systems -RecSys '13*, 185–192. https://doi.org/10.1145/2507157.2507171
- Hornick, M. F., & Tamayo, P. (2012). Extending recommender systems for disjoint user/item sets: The conference recommendation problem. *IEEE Transactions on Knowledge and Data Engineering*, 24(8), 1478–1490. https://doi.org/10.1109/TKDE.2011.90
- Dhanda, M., & Verma, V. (2016). Recommender System for Academic Literature with Incremental Dataset. *Procedia Computer Science*, *89*, 483–491. https://doi.org/10.1016/j.procs.2016.06.109
- Rohani, V. A., Kasirun, Z. M., Kumar, S., & Shamshirband, S. (2014). An effective recommender algorithm for cold-start problem in academic social networks. *Mathematical Problems in Engineering*, 2014. https://doi.org/10.1155/2014/123726
- Bogers, T., & Van Den Bosch, A. (2009). Collaborative and content-based filtering for item recommendation on social bookmarking websites. CEUR Workshop Proceedings (Vol. 532). https://doi.org/10.1007/978-0-387-85820-3
- Manouselis, N., & Verbert, K. (2013). Layered evaluation of multi-criteria collaborative filtering for scientific paper recommendation. *Procedia Computer Science*, 18, 1189–1197. https://doi.org/10.1016/j.procs.2013.05.285

Agency, H. J., Development, D., & Jeon, H. (2014). JCIT4150PPL, (March).

- Boukhris, I., & Ayachi, R. (2014). A novel personalized academic venue hybrid recommender. CINTI 2014 - 15th IEEE International Symposium on Computational Intelligence and Informatics, Proceedings, 465–470. https://doi.org/10.1109/CINTI.2014.7028720
- Sundermann, C. V., Domingues, M. A., Da Silva Conrado, M., & Rezende, S. O. (2016). Privileged contextual information for context-aware recommender systems. *Expert Systems with Applications*, 57, 139–158. https://doi.org/10.1016/j.eswa.2016.03.036
- Qiao, Z., Zhang, P., Cao, Y., Zhou, C., & Guo, L. (2014). Improving collaborative recommendation via location-based user-item subgroup. *Procedia Computer Science*, 29, 400–409. https://doi.org/10.1016/j.procs.2014.05.036
- Sun, Y., Barber, R., Gupta, M., Aggarwal, C. C., & Han, J. (2011). Co-author relationship prediction in heterogeneous bibliographic networks. *Proceedings -*2011 International Conference on Advances in Social Networks Analysis and Mining, ASONAM 2011, 121–128. https://doi.org/10.1109/ASONAM.2011.112

- Lao, N., & Cohen, W. W. (2010). Relational retrieval using a combination of pathconstrained random walks. *Machine Learning*, 81(1), 53–67. https://doi.org/10.1007/s10994-010-5205-8
- Asabere, N. Y., Xia, F., Member, S., Wang, W., Rodrigues, J. J. P. C., Member, S., Ma, J. (2014). Socially Aware Recommendation, 44(5), 689–700.
- Erdt, M., Fernández, A., & Rensing, C. (2015). Evaluating Recommender Systems for Technology Enhanced Learning: A Quantitative Survey. *IEEE Transactions on Learning Technologies*, 8(4), 326–344. https://doi.org/10.1109/TLT.2015.2438867
- Cantador, I., & Cremonesi, P. (2014). Cross-Domain Recommender Systems 2nd ed. of the RSs Handbook Cross-Domain Recommender Systems, (October).
- Fu, D., Liu, Q., Zhang, S., & Wang, J. (2016). The Undergraduate-Oriented Framework of MOOCs Recommender System. *Proceedings - 2015 International Symposium* on Educational Technology, ISET 2015, 115–119. https://doi.org/10.1109/ISET.2015.31
- Adomavicius, G., & Tuzhilin, A. (2008). Context-aware Recommender Systems. *Proceedings of the 2008 ACM Conference on Recommender Systems*, 335–336. https://doi.org/10.1145/1454008.1454068
- Burke, R., Felfernig, A., & Göker, M. H. (2011). Recommender Systems : An Overview, (1997), 13–18.
- Luong, H., Huynh, T., Gauch, S., Do, L., & Hoang, K. (2012). Using Author Network's Publication History. *Computer*, (1), 426–435.
- Lee, Y. (2015). Recommendation System Using Collaborative Filtering.
- Ochoa, X. ., Ternier, S. ., Parra, G. ., & Duval, E. (2006). A context-aware service oriented framework for finding, recommending and inserting learning objects. *Lecture Notes in Computer Science (Including Subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics)*, 4227 LNCS, 697–702. https://doi.org/10.1007/11876663_{75}
- Open Access proceedings solution for scientific conferences. (2016), 2016.
- Sezgin, E., & Ozkan, S. (2013). A systematic literature review on Health Recommender Systems. 2013 E-Health and Bioengineering Conference (EHB), 1–4. https://doi.org/10.1109/EHB.2013.6707249
- Amini, B., Ibrahim, R., Shahizan, M., & Ali, M. (2015). Expert Systems with Applications A reference ontology for profiling scholar's background knowledge in recommender systems. *Expert Systems with Applications*, 42(2), 913–928. https://doi.org/10.1016/j.eswa.2014.08.031

Urbanic, J. (2017). A Recommender System, 1-6.

Linden, G., Smith, B., & York, J. (2003). Amazon.com, (February).

- Ai-Lin Deng, Yang-Yong Zhu, B.-L. S. (2003). A Collaborative Filtering Recommendation Algorithm Based on Item Rating Prediction. *Journal of Software*, 14(9), 1621–1628. https://doi.org/10.13328/j.cnki.jos.2003.09.017
- Tsuji, K., Takizawa, N., Sato, S., Ikeuchi, U., Ikeuchi, A., Yoshikane, F., & Itsumura, H. (2014). Book Recommendation Based on Library Loan Records and Bibliographic Information. *Procedia - Social and Behavioral Sciences*, 147, 478– 486. https://doi.org/10.1016/j.sbspro.2014.07.142
- Conferences, M., & Website, M. C. (2016). SEARCH FOR CONFERENCES SEARCH BROWSE BY SPECIALITY, 1–2.
- Schein, A. I., Popescul, A., Ungar, L. H., & Pennock, D. M. (2002). Methods and metrics for cold-start recommendations. *Proceedings of the 25th Annual International ACM SIGIR Conference on Research and Development in Information Retrieval - SIGIR '02*, (August), 253. https://doi.org/10.1145/564376.564421
- Campochiaro, E., Casatta, R., Cremonesi, P., & Turrin, R. (2009). Do metrics make recommender algorithms? *Proceedings - International Conference on Advanced Information Networking and Applications, AINA*, 648–653. https://doi.org/10.1109/WAINA.2009.127
- Ekstrand, M. D. (2011). Collaborative Filtering Recommender Systems. *Foundations* and *Trends*® in *Human–Computer Interaction*, 4(2), 81–173. https://doi.org/10.1561/1100000009
- Pappas, N., & Popescu-Belis, A. (2014). Combining content with user preferences for non-fiction multimedia recommendation: a study on TED lectures. *Multimedia Tools and Applications*, 74(4), 1175–1197. https://doi.org/10.1007/s11042-013-1840-y
- Huang, T. H., & Huang, M. L. (2006). Analysis and visualization of co-authorship networks for understanding academic collaboration and knowledge domain of individual researchers. *Proceedings - Computer Graphics, Imaging and Visualisation: Techniques and Applications, CGIV'06, 2006*, 18–23. https://doi.org/10.1109/CGIV.2006.20
- Su, X., & Khoshgoftaar, T. M. (2009). A Survey of Collaborative Filtering Techniques. Advances in Artificial Intelligence, 2009(Section 3), 1–19. https://doi.org/10.1155/2009/421425
- R. J. Mooney and L. Roy, & Roy, R. J. M. and L. (1999). Content-based book recommendation using learning for text categorization. *Proceedings of the Fifth ACM Conference on Digital Libraries*, (June), 195–204. https://doi.org/10.1145/336597.336662

- Arnaboldi, V., Campana, M. G., Delmastro, F., & Pagani, E. (2017). A personalized recommender system for pervasive social networks. *Pervasive and Mobile Computing*, 36, 3–24. https://doi.org/10.1016/j.pmcj.2016.08.010
- Gong, S. (2012). Learning User Interest Model for Content-based Filtering in Personalized Recommendation System. *International Journal of Digital Content Technology and Its Applications (JDCTA)*, 6(11), 155–162. https://doi.org/10.4156/jdcta.vol6.issue11.20
- Giacomo, M. Di, & Mahoney, D. (2001). Mylibrary, a personalization service for digital library environments. *Personalisation*, 1–6. Retrieved from http://journal.mercubuana.ac.id/data/Giacomo.pdf

77